# Teacher Education F·o·r·m·i·n·g

# A Minnesola FRAMEWORK for Mathematics and Science

**Developed by** 

PATRICIA SIMPSON Associate Professor Biology St. Cloud State University

MARTHA WALLACE **Associate Professor Mathematics** St. Olaf College



**\*E**ducation in an academic discipline is no education at all if the student is not put to the intellectual labor of representing the discipline in lucid explanations, apt metaphors, and enlightening graphics; connecting the facts of the discipline to its underlying principles and to other bodies of knowledge; and applying the discipline to situations in life. Not only does such an education fail to serve future teachers, it fails all other students as well." (TYSON, 1994)



# ACKNOWLEDGEMENTS

This document was prepared by the Transforming Teacher Education Initiative Co-Directors Patricia Simpson and Martha Wallace in consultation with the Transforming Teacher Education Committee of SciMath<sup>MN</sup>. The authors extend their thanks to the committee members and others who worked long hours to provide information, offer suggestions, and critique earlier versions of this document. Special thanks go to committee member Dale Pearson, who collected and summarized much of the research that informed this document.

The authors also wish to recognize the efforts of Laurie Peterman and Anne Bartel from SciMath<sup>MN</sup>; and Department of Children, Families and Learning staff Sharon Stenglein and Kathleen Lundgren, who served as advisors to the project. Our work could not have been completed without the help of Debra Duffy, Donna Brakke, and Barbara Raymond who served as the coordinating staff for this project, and Mary Kay Peterson who assisted with editing.

Input into this document was provided by over 200 Minnesotans who attended Transforming Teacher Education conferences in 1994 and 1995. The document reflects comments and suggestions offered by the participants from these conferences, along with those who met again in July 1995 to critique the final document.

#### **Transforming Teacher Education Committee**

Mary Jo Aiken National Council of Teachers of Mathematics

Joel Burgeson University of Minnesota

Lisa Clemens, SciMath<sup>MN</sup>

Marcy Copeland Granite Falls/Clarkfield High School

Cyndy Crist, Chair, 1994–95 Minnesota State Colleges and Universities

Bruce Dickau College of St. Benedict/St. John's University

Rhonda Erdmann Willow Creek Middle School

Fred Finley University of Minnesota

John Frey, Co-Chair, 1993–94 Mankato State University

Russell Hobbie University of Minnesota Joyce Hummel Kennedy Elementary School

John Kemper University of St. Thomas

Larry Luck Anoka-Ramsey Community College

Dale Pearson Highland Park Senior High School

Judy Rohde John Glenn Middle School

Patricia Simpson St. Cloud State University

Martha Wallace St. Olaf College

Jim Whitney, Co-Chair, 1993–94 Honeywell Corporation

Cathy Wick St. Cloud State University

**BLANK SIDE** 



# PREFACE

This document offers a vision for what teacher education could be. It is intended as a guide for Minnesota post-secondary institutions as they consider the implications of national and state K–12 reform recommendations for teacher education programs.

SciMath<sup>MN</sup> is a public/private partnership founded in 1993 with a two-fold mission:

- 1. To be an advocate for standards-based, systemic reform for Minnesota's science and mathematics education system; and
- 2. To be a catalyst for implementation in Minnesota of the mathematics standards developed by the National Council of Teachers of Mathematics (NCTM), and of the science standards developed by the National Research Council (NRC).

To accomplish its mission, SciMath<sup>MN</sup> uses state legislative appropriations and private-sector funds to conduct initiatives in three broad programmatic areas: *policy* (changing state education policies to align with the vision of mathematics and science education in the national curriculum standards); *public awareness* (raising awareness of and support for standards-based reforms in mathematics and science education); and *professional development* (strengthening the preparation of new teachers and the continuing professional growth of practicing teachers in ways consistent with national mathematics and science standards).

In the areas of professional development, SciMath<sup>MN</sup> has launched several major statewide initiatives. To focus on issues related to pre–service preparation of mathematics and science teachers, SciMath<sup>MN</sup> organized a critique and consensus process which produced the following document. The document is thus the primary manifestation of the SciMath<sup>MN</sup> Transforming Teacher Education Initiative.

SciMath<sup>MN</sup> offers this document to the many stakeholders of Minnesota's teacher education programs, including not only institutions which are direct providers of teacher preparation courses, but also the K–12 constituencies of teachers, administrators, and policymakers as well as government, business, and community organizations interested in these issues.

Our hope is that this document will help engender and focus continuing dialog and consensus building around this emerging vision for teacher education in mathematics and science in Minnesota.

Bill Linder-Scholer Director of SciMath<sup>MN</sup>

begin the process of transforming teacher education in mathematics and science so that teachers will be prepared to teach according to the vision of present and future national standards and will be prepared to continue learning new content and new ways of teaching throughout their professional lives..."



# CONTENTS

#### Acknowledgements

#### Preface

Introduction: Transforming Teacher Education 1
A Minnesota Framework for Mathematics and Science
Creation of the Document
Purpose of the Document
Audience for the Document
Overview of the Document
Scope of the Document
Assumptions of the Document
Terms Used in Special Ways in the Document
1. Context for Change 5
A Call for Change
Changing Curriculum
Changing Instruction
Changing Teachers' Knowledge
Changing Teacher Education Programs
The Old Paradigm
The Need for Change in Mathematics and Science Education
New Standards
A New Vision of Students
A New Vision of Teachers
A New Vision of Teacher Education
Teacher Education at Riverview State University: A Scenario
2. Standards for the Education of Teachers of
Mathematics and Science 19
SECTION I: Knowing Mathematics and Science
SECTION II: Knowing Pedagogy in Mathematics and Science
SECTION III: Knowing Students as Learners in Mathematics and Science
SECTION IV: Establishing an Environment for Learning
Mathematics and Science
SECTION V: Developing as a Teacher in Mathematics and Science

of Teachers of Mathematics and Science	43
SECTION I: What Should Faculty do to Educate Accomplished Teachers?	
SECTION II: What Should Institutions do to Educate Accomplished Teachers?	
SECTION III: How Should Society Provide Support for the Education of Accomplished Teachers?	
4. Transforming Policy to Practice	_ 51
The Time for Change	
Challenges to Meet	
An Organizer for Reform	
Why Policy is Not Enough	
A Call to Action	
Conclusion	
References	5
Appendices	5
APPENDIX A: National Standards for Mathematics and Science	
Teaching and Teacher Education	
APPENDIX B: National Standards for Curriculum, Instruction, and	
Assessment in Mathematics and Science	
APPENDIX C: The History of Transforming Teacher Education and the Role of SciMath <sup>MI</sup>	1
APPENDIX D: Mathematics and Science Bibliographies	
APPENDIX E: List of Participants	



# Introduction

**Transforming Teacher Education** 

### A MINNESOTA FRAMEWORK FOR MA THEMATICS AND SCIENCE

This document advocates a Minnesota vision for transforming teacher education in mathematics and science. In this vision:

Beginning mathematics and science teachers will be prepared to teach according to the vision of present and future national Standards, and they will be prepared to continue learning new content and new ways of teaching throughout their professional lives.

It is important to note that a vision is a picture of the future lying just beyond the grasp of today's reality. A shared vision is necessary to provide all stakeholders with a clear picture of the direction for change. Realizing the vision will require imagination and creativity on the part of those involved in the process, and a recognition that not all institutions will attain the vision in the same way. Alternative methods are necessary to achieve the vision in light of the differing needs, realities, and goals of Minnesota teacher education institutions.

#### Creation of the Document

This document is a response to national Standards for mathematics and science education, as well as to the emerging Minnesota Graduation Standards for K-12 students and pending changes in Minnesota's teacher licensure system. It is a product of SciMath<sup>MN</sup> through its Transforming Teacher Education Initiative. This is a first attempt to delineate the Minnesota vision. It is important to note that the sense of the vision will evolve over time as the needs of beginning teachers change.

distinct sources. Initial input was provided by Minnesotans directly involved in K-16 mathematics and science education who attended several state-wide meetings to determine the knowledge and skills needed by teachers of mathematics and science. Additional information was gathered through a review of the existing research literature related to student learning, effective teaching, and teacher education. The final information source involved the analysis and synthesis of national and state Standards for mathematics and science education, along with other reports providing recommendations for teacher education. Throughout the evolution of the document, the ideas presented were reviewed and evaluated by stakeholders in teacher education.

#### **Purpose of the Document**

The purpose of this document is to:

- 1. Stimulate and inform state dialogue among all stakeholders of teacher education:
- 2. Guide Minnesota institutions of higher education as they consider their roles and responsibilities in mathematics and science education reform; and,
- 3. Inform the development and implementation of state licensure rules for teachers of mathematics and science.

The resulting document is intended as a tool for discussions of what teacher education programs in Minnesota could be. It is not a checklist of mandates.

expected of our teaching force now than ever before. Successful efforts to restructure schools for the demands of a knowledge-based economy depend critically on the nation's teachers."

(INTASC, 1995)



#### Audience for the Document

This document is intended for distribution to and discussion among all individuals involved in planning and providing the education of K–12 mathematics and science teachers. This list includes students involved in teacher education programs; faculty in mathematics, science, and education departments; field experience supervisors and cooperating teachers; university and K–12 mentors; leaders of professional organizations in mathematics and science; university and school administrators; and representatives from business, industry, government, and funding agencies.

#### **Overview of the Document**

The four chapters in this document describe recent reform initiatives in mathematics and science education, delineate the vision for transforming teacher education in Minnesota, make recommendations for systemic change, and discuss the steps necessary to transform teacher education policy into practice.

CHAPTER 1 sets the context for the document by describing the current status of science and mathematics education and presenting a vision for the future. It outlines the background of recent and emerging national *Standards* for curriculum, instruction, and teacher education in K–12 mathematics and science, and identifies the special conditions affecting mathematics and science education in Minnesota. Chapter 1 concludes with a scenario that presents an imaginary university which has aligned itself with national and state *Standards*.

CHAPTER 2 presents a framework for the Minnesota vision. It summarizes and synthesizes standards for beginning teachers' knowledge and skills, and describes

experiences that enable them to acquire those skills and knowledge. It then lists necessary roles and responsibilities of faculty involved in all aspects of mathematics and science teacher education, along with characteristics of institutions that produce accomplished beginning teachers.

CHAPTER 3 describes the support needed by Minnesota teacher education programs to effect the transformation of teacher education in mathematics and science. It provides recommendations for various stakeholders

in teacher education including faculty, institutions, Minnesotans, government, business, industry, professional organizations, K–12 schools, and school districts.

CHAPTER 4 discusses the challenges to be met if Minnesota is to transform the policy presented in this document into practice. It provides an organizer for use in examining the process of reform, and presents a call for action to all of those involved in teacher education.

#### Scope of the Document

Changing the knowledge base for mathematics and science teachers requires two concurrent efforts: reform in the education of beginning teachers, and reform in the professional development of current teachers. However,

the vision and framework presented in this document focus only on the education of beginning teachers of mathematics and science, leaving the continuing education and professional development of current teachers to future initiatives.

This document refrains from making recommendations to institutions regarding the content of specific courses, majors, or programs. Instead, it



provides a resource with which institutions of higher education can examine their teacher education programs,

and develop specific plans to revitalize them. These plans must include consideration of each institution's unique identity, its mission and goals for teacher education, and the specific needs of its students.

This document does not attempt to address all aspects of teacher education. Its focus is on those aspects that are unique to mathematics and science or that are addressed in national and state *Standards* for these disciplines. Other equally necessary, but more general, aspects of teacher education are not included.

#### Assumptions of the Document

This document is built upon the following assumptions:

- 1. A teacher is a decision-making professional who can recognize and generate options when faced with problems.
- 2. Teachers are the key to effecting mathe-matics and science literacy for all students.
- 3. The education of mathematics and science teachers is the responsibility of the entire institution.
- 4. Successful transformation of teacher education requires strong and sustained support from all stakeholders.
- Effective K–12 teaching requires a continuum of professional development experiences that allow mathematics and science teachers to continue

learning throughout their career.

- Successful efforts to transform the education of mathematics and science teachers will enhance the education of every student enrolled in an institution's mathematics or science class.
- 7. All students in mathematics or science classes are potential teachers.

# Terms Used in Special Ways in this Document

This document makes careful use of words and phrases that may have different meanings to various members of the teacher education community. In particular, note the following definitions:

ACCOMPLISHED TEACHERS: Those who have expertise and skills, appropriate for their level of experience, that enable them to teach according to the vision of national and state *Standards*.

BEGINNING TEACHERS: Those whose level of experience appears somewhere on a continuum from entry into student teaching through their first years of teaching. The distinction between beginning and advanced teachers is not in the types of knowledge and skills they possess, but rather in the degree of



Introduction

sophistication they exhibit in the application of those skills and knowledge (INTASC, 1995).

DISCIPLINE: The subject area of a teacher's expertise. In the case of non-specialist elementary and middle school teachers, this means a broad knowledge of mathematics and the sciences. In the case of secondary teachers, the term discipline refers to a deep under-standing of whatever subject(s) they will be teaching, typically mathematics or one or more of the sciences, combined with a broad view of mathematics and science as related disciplines.

INSTITUTION: A college or university that educates teachers of mathematics and science.

STANDARDS, NATIONAL: The aggregate of recommendations by national mathematics, science, and/or teacher education professional organizations for curriculum, instruction, assessment, and the education of teachers in mathematics and science. An annotated list of the *Standards* documents synthesized here appears in Appendices A and B.

STANDARDS, STATE: *Minnesota's Graduation Standards*, particularly the *Profile of Learning*, which includes content standards for mathematics and science. This document references the June 1995 draft of the *Profile of Learning*.

STUDENTS, ALL: Every K–12 student, regardless of ability, career goal, gender, ethnicity, or socioeconomic status, who is enrolled in a mathematics or science class. TEACHERS OF MATHEMATICS AND SCIENCE: All teachers who assume responsibility for the mathematics and/or science development of children in grades K–12.



Introduction



Chapter I

# Context

for Change

# A CALL FOR CHANGE

he reform of teacher education can be considered only in relation to some vision of what it ought to be. ... The yardstick is whether the children taught by graduates of teacher education programs can understand as well as memorize, apply as well as state. imagine as well as copy, solve problems rather than shrug them off, and make themselves felt in a society that seems to be in trouble for lack of these capacities."

(Tyson, 1994)

As we enter the last half decade of the 20thcentury, the call for change in school mathematics and science is clear and pervasive. A call for mathematical and scientific literacy for all has been a common message in almost every report that describes the status of our current education system, and the needs of students who will be successful in the future. In order to achieve the intent of this message, nearly every major professional organization in mathematics and science has published new standards which call for reform in three critical dimensions of mathematics and science education: curriculum, instruction, and teachers' knowledge. Reform in these three dimensions requires a concurrent change in teacher education programs.

#### **Changing Curriculum**

K–12 curriculum, the first dimension, describes what should be taught in mathematics and science classrooms.

In describing curriculum, reports agree that it is not possible to teach all that is known about either mathematics or science. Instead, deep knowledge of fewer topics is required for literacy with an emphasis on deep understanding of the major themes of the discipline. An important part of this deep under-standing involves making connections between the discipline and the world beyond the classroom. Skills for exploring new problems and meeting the challenges of the future are also emphasized. Curriculum materials focus on the dynamic nature of the disciplines, and the role of technology in learning and doing mathematics and science.

#### **Changing Instruction**

The process of K–12 instruction, how content is taught and assessed, is the second important dimension of reform. Influenced by research on cognition, national Standards emphasize that how the content is taught is at least as important as what content is taught. If students are to view mathematics and science as an integrated whole instead of a fragmented collection of arbitrary topics, the study of mathematics and science should be consistent with the nature of inquiry in the discipline. Instruction must stress student actions such as exploring, investigating, discussing, analyzing, constructing, hypothesizing, and validating. Rather than focusing on memorization and competition, instruction according to the Standards encourages students to guestion and think creatively, often as part of a team, about solutions to problems of personal interest or importance to the discipline. Teachers must structure learning so that students can learn habits for inquiry and problem solving at the same time as they develop personal knowledge of the discipline.

#### **Changing Teachers' Knowledge**

The substantive changes called for in K–12 mathematics and science curriculum and instruction require concomitant changes in the knowledge base needed by teachers. This is the third critical dimension of mathematics and science education reform. In order to guide students in active inquiry and discovery, teachers must possess a deeper understanding than ever before in four areas: knowledge of their discipline, knowledge of pedagogy, know-ledge of students as learners, and knowledge of the learning and teaching environments in which they will work. In addition, teachers must also understand the nature of quality professional development opportunities, and possess dispositions for continuing professional growth and development.

#### **Changing Teacher Education Programs**

To achieve the necessary teacher knowledge base requires a new look at the education of beginning teachers of mathematics and science and their continuing professional development. Minnesota is currently developing a new teacher licensure system that emphasizes teachers' knowledge and competencies instead of the present course-based system. Both national and state changes challenge Minnesota higher education institutions to transform their teacher education programs. By aligning teacher education programs with state and national *Standards*, colleges and universities can take advantage of the opportunity to affect meaningful reform in K–12 mathematics and science.

This framework provides a tool with which teacher education institutions may examine and renew their programs. To set the stage for the process, this chapter contrasts the old paradigm of mathematics and science education with the new vision presented by the *Standards*.

#### The Old Paradigm

Prior to the 80's, school mathematics and science education in the United States and in Minnesota primarily prepared students to function effectively in the social and political climate of the time. Most students learned "shopkeeper" arithmetic and "encyclopedic" or "survey" science. Advanced mathematics and science students learned, in rote fashion, the content that would prepare them for college courses. Mathematics and science were taught as a series of facts, terms, formulas, and procedures with few connections. Neither mathematics or science used extensive problem solving nor decision-making. Science courses might include a few laboratory experiences, but even these experiences only gave students practice in verifying information the instructor or the textbook had already presented.

In mathematics and science courses, as in all other school subjects, students "learned" by watching and listening, copying and drilling. Some students, who were considered "good at" mathematics or science, were encouraged to take more courses on topics of special interest. Others were tracked into "low-ability" groups or courses with uninteresting material and low expectations. Only college-bound students took more than one year of mathematics or science in high school, and only the future science or mathematics majors took trigonometry, chemistry, physics, or advanced biology.

The role of the teacher was prescribed by the curriculum and the "ability level" of the class. Teachers of mathematics and science, at all grade levels, were expected to transmit an established body of knowledge to a homogeneous and passive audience. Lesson planning consisted primarily of making the textbook content palatable to the students, selecting worksheets, and deciding how much homework to assign. High school and junior high school science teachers were also responsible for selecting demonstrations or experiments to illustrate the principles being taught. Teachers had to be masters of the material they covered, but not much else.



Elementary teachers were expected to teach mathematics and science to their students even though most had only a year or two of high school mathematics or science and no college coursework in either discipline.

A brief flurry of reform in mathematics and science during the 1960's produced meaningful curriculum with goals related to the structure of the disciplines and their processes of inquiry. The teacher was still viewed as the authority, and as a result professional development programs for teachers often focused largely

on enhancing content knowledge. High school mathematics and science teachers had many opportunities to attend National Science Foundation (NSF) summer and year-long institutes on advanced concepts in their discipline. But these NSF institutes paid little

or no attention to the knowledge and skills teachers needed to translate a teacher's personal knowledge into enhanced student understanding. New curricula were developed for students in the elementary grades, but elementary teachers were given little help in implementing new curricula, and most schools provided minimal budgets for supplies. Despite the efforts of these programs, an extensive study of teaching conditions published in 1981 still found three conditions to be most evident: the textbook was the basis for instruction, the teacher determined the tone and type of learning experience, and lecture-discussion was the prevalent mode of presentation (Harms and Yager:1981).

Some students were successful under the old paradigm (e.g., the authors and readers of this document). But a great majority were not successful and opted out of mathematics and science as soon as possible. Mathematics and science were considered subjects for the elite, and it was socially acceptable to admit to being "mathematically phobic" or "scientifically illiterate."

### The Need for Change in Mathematics and Science Education

Dissatisfaction with K–12 education resurfaced in the 1980's. Beginning with the release of *A Nation at Risk* in 1983, one national report after another decried the current state of American education, particularly in comparison with industrially competitive nations like Japan. At the same time, new research paradigms were generating new theories on how children learned, and new technologies were exploding the available knowledge base and the need for scientifically literate workers. Mathematics and science led the other disciplines in developing national standards

or recommendations for school learning. The term standards, in most of the reform documents, referred to "visions of performance and its associated knowledge and skills" (Richardson:1994:16), not as "goals and their assessments" (Richardson:1994:16), as the term is often used in state graduation or licensure documents.

A catalyst for the new *Standards* for K–12 student learning was the realization that the current system was not producing citizens who could compete in the world marketplace. Other factors included:

- New national goals calling for mathematics and science literacy for all, not just for a few mathematicians and scientists.
- Research results that highlight the necessity for new ways of teaching.
- A rapid explosion of knowledge, leading to the realization that fact-based education was no longer possible or sufficient.

- The increasing impact of technology on learning and doing mathematics and science.
- The enhanced role of mathematics and science in everyday life.

In addition, educational policies in Minnesota were also affected by socioeconomic factors. The population of the state has become more urban and culturally diverse. The economy has changed from one with an agricultural and mining base to one focused on growing markets for service, both local and international. Families with a single parent or two working parents are now the norm. The number of children in Minnesota classrooms with special needs has increased, as has the level of stress faced by many children.

#### **New Standards**

Since 1989, an abundance of educational reform documents have been published, each referred to as "standards." Both the National Council of Teachers of Mathematics' Standards for Curriculum and Assessment in School Mathematics and the American Association for the Advancement of Science's Project 2061: Science for All Americans were released in 1989, followed by documents from many other organizations. Professional mathematics and science organizations that have issued national standards for K–12 curriculum, instruction, and assessment include the American Association for the Advancement of Science (AAAS), the National Council of Teachers of Mathematics (NCTM), the National Research Council (NRC), and the National Science Teachers Association (NSTA).

These same organizations, along with the Mathematical Association of America (MAA), National Board for Professional Teaching Standards (NBOT), and the Interstate New Teacher Assessment and Support Consortium (INTASC) have also issued standards for mathematics and science teaching and teacher education. Together these organizations have created a new vision for students, teachers, and teacher education.

A complete list of these organizations and their reports can be found in Appendices A and B. A summary of common themes in the reform documents is presented in Table 1.

#### A New Vision of Students

These documents create a vision of new students and new ways of learning mathematics and science. In this vision, mathematics and science are for all students. These students experiment, inquire, hypothesize, generalize, organize, and evaluate. They solve problems presented by the teacher, but, more importantly, they generate their own guestions and problems from personal observations of phenomena. They are familiar with technology and other tools, and know how and when to use them appropriately. They see themselves and their peers as sources of knowledge, and do not rely totally on the teacher to see if what they are doing is "right." They read and communicate orally and in writing about mathematics and science, and they use mathematics and science to understand or explain other parts of the world. They have the confidence and the skills to continue learning, with or without a teacher.

In this new vision of students, mathematics and science are about ideas, not just about procedures. Further, they are about big ideas and connections, not just isolated facts and concepts. Knowing mathematics and science is doing mathematics and science.



### **TABLE 1 - COMMON RECOMMENDATIONS OF REFORM DOCUMENTS**

#### STUDENT RECOMMENDA TIONS

- Mathematical and scientific literacy for ALL students
- Active, hands-on learning
- Critical thinking and problem solving
- In-depth study of fewer topics
- Conceptual learning and understanding
- Cooperative and collaborative activities
- Connections within and across disciplines
- Full use of appropriate technology for learning
- Authentic assessment aligned with instruction
- Life-long learning

### **TEACHER RECOMMENDA TIONS**

- Teachers as facilitators, catalysts, and coaches
- Teachers as change agents
- Teacher involvement in decision making
- More team teaching
- Improved teacher preparation
- Use of exemplary teachers as mentors and role models
- Use of multiple types of assessments of teaching
- Emphasis on continuing professional development



9

#### A New Vision of Teachers

Teachers in this new vision know the futility of trying to pour knowledge into seemingly empty young heads. Instead, they facilitate and guide their students' learning by helping them to integrate new and previous knowledge.

They serve as coaches and catalysts for learning. They empower their students to become independent learners, to question

and evaluate their own and their classmates' conclusions without always referring to the teacher or the textbook for "the truth."

These teachers present information within relevant contexts, rejecting large amounts of drill and the memorization of unconnected facts. They identify, and help students identify, misconceptions that may hamper their

learning of new ideas. They make full use of appropriate technology and manipulative models as tools for teaching, learning, and doing mathematics and science. They continually question and rarely tell. They

help students learn to work cooperatively, communicate clearly, and reason deeply.

When planning for instruction, teachers identify the big ideas and overarching

themes of the content they are teaching, and develop

ways to help their students see the connections. They generate multiple strategies to represent the content to accommodate diverse learners and learning styles. They expect to cover less content, but to "uncover" it more deeply. They plan questions and experiences, not lectures. They develop multiple methods to assess students' learning in ways that are authentic to the content and aligned with instruction, and they use assessment results to improve both student learning and their own teaching.

In the new vision, teachers recognize that teaching is a complex practice, not reducible to prescriptions or recipes. They understand mathematics and science as dynamic disciplines, and realize that they must continue to learn about their subject, their students, and instructional strategies throughout their professional lives.

#### A New Vision of Teacher Education

Teachers of mathematics and science at any level of the K–12 continuum need different knowledge and skills than before to enable them to teach according to the new visions of students and teachers. Thus the national standards imply a new vision for the education of mathematics and science teachers. The remainder of this document describes in detail the vision of teacher education implicit in the national standards and informed by current reform efforts in Minnesota. The scenario which follows describes one possible vision of teacher education.





#### TEACHER EDUCATION AT RIVERVIEW STATE UNIVERSITY

#### A Scenario

Riverview State University (RSU) has a national reputation for teacher education. One of the reasons for the institution's quality reputation is its continuing emphasis on evaluation and renovation of programs to ensure that its graduates are prepared to teach not just in Minnesota's schools of today, but also the schools of tomorrow. The university demonstrated this commitment to quality teacher education by being one of the first institutions in Minnesota to join Project Renaissance, a national association

of liberal arts and undergraduate institutions committed to exemplary programs for teacher education. Effective teaching at all levels is the number one goal of the institution.

The largely rural area in which the university is located is just now beginning to develop the ethnic diversity that is common in most college communities of comparable size. The university community is more diverse than it once was, partially due to the changing demographics of the state and also as a result of an intensive recruiting program undertaken by the university. The university has worked to increase the diversity of its student and faculty community to more clearly reflect the demographics of the areas in which its graduates will work. In teacher education, the university has strengthened its ties to Twin City schools, and now offers a special emphasis for students who want to teach in urban schools.

RSU's success in teacher preparation is also due to its emphasis on collaboration and innovation in teaching throughout the institution. Faculty and administrators from various departments and colleges routinely work in teams for planning and instruction. Their efforts are recognized and rewarded in many ways by the institution. Time

is allocated in faculty loads for this collaboration. At

promotion and tenure reviews, faculty are recognized for their contributions to enhance teaching in the same manner that others are recognized for their contributions through research and grantsmanship. Risk taking in teaching is also rewarded. Faculty are encouraged to implement new ideas in curriculum, instruction and assessment. The "failure" of a new idea in teaching is seen as an opportunity for professional growth and development. Respect and trust among faculty and administrators across campus is the norm, and they share a common belief that all departments are responsible for the successes and failures of the teachers who graduate from Riverview.

#### The Teacher Education Program

The resulting program is one in which all participants take pride. Teachers who graduate from Riverview have strong backgrounds in content and pedagogy, extensive opportunities

for clinical experience in a variety of settings throughout the program, and continuing support as they begin their first years of teaching. During their first years at RSU, all students complete solid general education courses in science and mathematics that help them develop a conceptual understanding of the nature of the disciplines, focus on general concepts, and build connections between the disciplines and the world at large. Through strong partnerships with local community colleges, transfer students are able to benefit from the same sorts of quality coursework in content and general education courses. Teacher education majors formally enter the program no later than the beginning of the sophomore year. Throughout the rest of their undergraduate program, courses that focus on pedagogy are paired with content courses and actual school experience. Groups of students enter the program and function as cadres over the next three years. Advising is done by a team of faculty who

continue to work with students during their first years of teaching. Applied psychology and multicultural education courses still exist, but they are tailored to the disciplines and age levels at which teacher education majors will be working. Similar technology and exemplary curricula are utilized in all courses and in area schools. Faculty share common assumptions about the impact of student diversity on learning. All faculty understand that meaningful learning requires making connections between new and existing knowledge. They have knowledge of the experiences that students receive in their other classes, and are able to make use of this information in helping students reflect on what they are learning and how it might be useful in future teaching.

Clinical experiences are a clearly defined part of the teacher education program at Riverview. The courses are taken in a sequence that begins formally in the sophomore year. Students and cooperating teachers receive a list of outcomes to be achieved through their clinical experiences. At first, students work largely with tasks that focus on individuals or small groups, and then as their time and experience in the classroom grows, so does the number and diversity of the students they are assigned to work with. Clinical experiences are always tied to a class in the program, or to the bi-weekly seminars held at the RSU Mathematics and Science Center. The seminars present ideas that link field experience to research on topics such as curriculum and instruction innovations, new assessment techniques, or learning theory. Commonly the audience for a seminar may include discipline-based and education faculty from the university and local schools, students of all ages, and even recent graduates of the program. By the senior year, students have well in excess of 200 hours in the schools even before their actual student teaching experience begins.

A variety of nonformal student experiences also exist. As firstyear students, many begin jobs in schools as lunchroom, playground, or instructional aides that they continue to hold throughout their college career. A significant number of junior and senior elementary science and mathematics majors run an afterschool enrichment program in the local school district called School Stop. Other teacher education science and mathematics majors are involved with Scientific Discovery, a series of residential science and mathematics programs

that the university runs each summer for under-represented K–12 students.

As seniors, students begin the year in the schools at which they will be student teaching. After participating in beginning-of-the-year activities at the school, they return to campus for a formal science or mathematics methods course in which they work to develop the units they will later teach in the schools. This arrangement gives methods courses greater meaning because students can see the direct application of their efforts. After four weeks on campus, they return to the school and student teach for the rest of the semester. During student teaching, they are observed by faculty from mathematics and science as well as from education departments, and they continue to meet with the on-campus seminar team either in person or electronically.

Student teaching is always scheduled so that the students spend their final semester back on campus. During the last semester, they are involved in a series of capstone courses designed to help them process their experiences and internalize the attitude that teaching science or mathematics involves a life-long commitment to professional development and learning.

Teachers who provide or supervise clinical experiences for Riverview students are identified through an application process. Summer professional development programs for local teachers, the SciMath<sup>MN</sup> Teacher Academy, and the Department

of Children, Families and Learning Best Practice Networks have helped RSU identify teachers to work with students in field experiences. These teachers complete an introductory course on supervision, and are then given opportunities to



choose the sort of involvement they would like to have based on available positions. The teachers who participate receive free university credit for serving as field supervisors, and many have used these credits to enhance their own professional development.

# Exemplary Programs in Mathematics and Science

Mathematics and science teacher education at all levels have been especially strong programs at RSU for more than twenty years. These excellent programs benefit from the institution's commitment to quality teaching, but also from the number, diversity, and collaborative efforts of the faculty from both the College of Education and the College of Arts and Sciences. The College of Education has faculty with special expertise in mathematics and science, while each of the mathematics and science departments contain faculty with extensive K–12 classroom teaching experience. These "teaching experts" within the content departments teach special content courses for future teachers, the secondary teaching methods courses, and assist with supervision of student teachers. They also work to enhance the effectiveness of instruction

for all students enrolled in the departments' mathematics and science courses. This large cadre of faculty interested in mathematics and science education has made it easier for the university to respond to the challenge of new K–12 state and national standards for science and mathematics education, and the opportunity provided by new state teacher licensure programs.

Resources at the institution are limited and faculty teams have had to be creative in finding ways to fund their innovative programs. Faculty from RSU are active in state and national projects, and this activity has allowed the university to participate in

a number of state and federal initiatives. These initiatives have helped Riverview to garner resources in science and mathematics that would not otherwise be available. Federal monies were used by the institution to purchase materials and establish the Mathematics and Science Education Center with two computer laboratories, model classrooms for mathematics and science instruction, and an extensive curriculum resource room. State monies provide many of the resources required to fund professional development opportunities for teachers and summer programs for K–12 students. Private funding has been used to support teaching fellows on campus. These fellows are classroom teachers chosen from local schools throughout the state.

#### **Technology Resources**

Interactive TV laboratories in the Mathematics and Science Center allow RSU students to observe school and classroom settings without leaving the university, provide professional development classes for teachers at distant schools, and allow

for interactive seminars with students at each professional development site. During their student teaching practicum, students are provided with a lap-top computer and software, access to the Internet for interactions with each other and with the methods classes on campus.

The curriculum and computer resource rooms serve as major sites for the instruction of preservice teachers. In the science and mathematics curriculum room, students have opportunities to evaluate and utilize the latest teaching materials. The room contains complete sets of new curriculum materials along with an extensive collection of print materials including textbooks. Journals focusing both on research and classroom activities which report on current research and recommendations from professional organizations in mathematics and science are also available. Videos and laser discs

are available in the curriculum room, while the best software is found in the nearby computer labs.

Other available classroom aids include calculators, manipulatives, microscopes, models, telescopes, celestial globes, air track tables, and almost any other tool (from a rock hammer to a dissecting kit) that students might need in



a well-equipped science or mathematics classroom.

The computer classrooms are equally well equipped. Each room has a complete set of 24 computers available with Internet access and CD Rom. Exemplary instructional software is available at all grade levels in both mathematics and science, along with software packages for word processing, spreadsheets, and graphing. Teacher tool kits that include grade programs and classroom organizers are also located here. Probeware is available for use in both the mathematics and science labs, and students can learn to utilize these tools in either the elementary or secondary Mathematics, Science, and Technology courses. It is in these classes that they also get an opportunity to use HyperCard to build stacks for lessons which include real time data as well as images taken from the Internet or laser discs.

As with all RSU students, mathematics and science education majors have a basic familiarity with the computer. They are required to take an introduction to computing course that covers Internet, word pro-cessing, spreadsheets, and graphing programs.

They use the computer to collect and analyze data in mathematics and science content courses and perform library searches. This means that the technology courses in teacher education have finally reached the point at which they can be used to focus on appropriate use of technology for instruction rather than learning basic skills.

The curriculum and computer room supervisors are local teachers on leave from their schools. They are serving this year as Riverview Eisenhower fellows. As a part of their jobs, each semester they team teach one course for preservice teachers with an

RSU faculty. They are each a part of a seminar team presenting special topics in science and/or mathe-matics education, and they spend the remainder of their time in the curriculum or computer rooms. This time gives them an opportunity to learn about new curriculum, and also provides a strong resource for preservice teachers.

#### The Process of Change

Changes in programs and courses at the university did not take place overnight. Through the years, barriers had been established between the university and local schools, as well as between faculty in the various colleges and departments. It seemed that conversations between various stakeholders in teacher education more often focused on placing blame than on discussing the needs of students and collaboration. The first steps toward reform involved extensive dialogues among University planning teams to develop a shared vision of what teacher education graduates should know and be able to do, as well

as the roles and responsibilities of each group in achieving this goal. This process was an important first step in reestablishing trust and respect among all of those involved. During this discussion, each group's contributions and needs were uncovered. Education faculty were able to provide important information about educational research and the application of various methodologies and curriculum, but they needed help in understanding the prior knowledge base of university students in the disciplines. Mathematics and science faculty could provide this information, and they were also able to identify those concepts which are most difficult for students. What many of them lacked was knowledge of present day school environments. Classroom teachers provided this knowledge, but needed help to understand how the structure and constraints of the university could challenge the implementation of ideas that would enhance the quality of RSU graduates. Background knowledge and current experience with new reform documents varied widely among individuals in each of these groups. Administrators served as important facilitators of the entire process by providing support and stimulus as needed.

#### **Connections with Schools**

RSU currently has strong connections with the surrounding region's elementary, middle, and high schools. A number of years ago the university became a founding partner of their locally developed Coalition for Better Science and



Mathematics Education. This program established partnerships between the university, local schools, and business in an attempt to improve student achievement from kindergarten through college by all studying mathematics and science.

Coalition members make decisions based on the premise that education is a continuum from preschool through university, and that improvement in one part of the system is not possible without improvement in all parts. The purpose of the partnerships then is to enable all members of the Coalition to reform both preservice and inservice teacher education while simultaneously reforming local schools. All partners share a common goal of producing students and teachers who are life-long learners willing and able to utilize mathematical

and scientific knowledge and skills in daily life.

The Coalition began the process of change at the K-12 level by examining emerging state and national recommendations for quality science and mathematics education, and the research related to effective teaching and learning. They examined existing programs and the needs of local students, and used this information to revise student outcomes in science and mathematics. Then teams of university faculty, local teachers, and community members joined to examine exemplary programs in school science and mathematics. They also examined and evaluated curriculum resources for adoption by the local schools. During the next few years, business worked with the schools and university to help acquire the technology and curricular resources needed to implement the new programs. The university made a commitment to obtain similar materials, and incorporate them as appropriate into existing university courses, to promote a seamless transition for preservice and inservice teachers who enrolled in local courses and then worked in area schools. Business helped by identifying local sites to provide teacher and

student intern programs in addition to their contribution of financial support and personnel. Throughout the process, an exchange program was used to share the expertise of all participants.

University faculty were paired with individual schools, and began to spend two to three days a week in these schools. This not only included the time they spent supervising clinical experiences,

but also allowed them to teach classes, provide professional development activities, and work with teachers to bring additional resources into the schools.

In exchange, teachers in mathematics and science became more involved with teacher education programs, and some teachers were given formal appointments at the university. Teacher contributions included supervising clinical experiences, serving as mentor teachers for student teachers, and working with the university's mentoring programs for beginning teachers. They also worked in the university computer and curriculum rooms, and team taught both content and methods courses at the university. The partnerships and shared roles developed in this project continue today.

#### **Change Continues**

The Coalition partnerships that were formed to simultaneously renovate the Riverview teacher education program and local K–12 education in mathematics and science continue to examine and refine their successful model. Planning teams are composed of education specialists (science and mathematics educators, and curriculum and instruction generalists), school system practitioners (teachers and administrators), content specialists (scientists and mathematicians), and university students. The planning teams were responsible for developing the initial program requirements, and turning these requirements into preservice courses and clinical field experiences. They continue to oversee the implementation,



ongoing evaluation, and revision of the program. Indeed, at Riverview State University, program improvement is both a reality and a continuing goal.



[BLANK PAGE]



# Chapter 2 Standards

for the Education of Teachers of Mathematics and Science

### CHAPTER OVER VIEW

**The education** of teachers consists of all of the life-long experiences, inside and outside of classrooms, that contribute to the knowledge, skills, beliefs, and attitudes that a teacher brings to the teaching situation."

(POETS, 1994)

This chapter describes a Minnesota vision for Transforming Teacher Education. It presents standards for what beginning mathematics and science teachers should know and subsequently be able to do. It identifies the experiences which lead to the development of those skills and knowledge, and describes faculty and institutional roles and responsibilities in providing these experiences.

The chapter is divided into five sections: Section I Knowing Mathematics

- and Science
- Section II Knowing Pedagogy in Mathematics and Science
- Section III Knowing Students as Learners of Mathematics and Science
- Section IV Establishing an Environment for Learning Mathematics and Science
- Section V Developing as a Teacher in Mathematics and Science

Each section is further divided into four subheadings that provide answers to these

#### questions:

- A. What must teachers know and be able to do?
- B. What experiences should teachers have to develop their knowledge and skills?
- C. What are the roles and responsibilities of faculty in helping teachers develop their knowledge and skills?
- D. What are the characteristics of institutions that produce accomplished teachers?

#### **Using this Chapter**

This chapter is designed to serve as a basis for dialogue among many diverse stakeholders in the teacher education community. With this purpose in mind, it is appropriate to apportion the responsibilities for careful reading of specific sections among several individuals. This reading can then be followed by cooperative interaction and joint planning by all stakeholders at each teacher education institution. Based on the assumption of multiple readers and cooperative discussion, there is some duplication among sections. Content knowledge is an essential component of an individual's preparation as a teacher of science and mathematics. Teachers need a "deep knowledge" of the discipline in order to teach science or mathematics according to the vision of the national and state *Standards*. Harriet Tyson, the author of *Who Will Teach the Children?*, describes this knowledge:

"The 'deep knowledge' teachers need most is the ability to answer the 'why' and 'what for' questions that their students ask, in addition to knowing the basics of the field of study. We need teachers, at every level, who recognize and explicitly deal with the obstacles students face as they struggle to understand, articulate, and apply new knowledge."

(Tyson, 1994)

In this section, the "deep knowledge" of mathematics and science required of beginning teachers is organized into seven categories.

Beginning teachers will:

- 1. Understand the fundamentals of their discipline.
- 2. Understand the nature of their discipline from a broader perspective.
- 3. Apply the discipline in their own lives.
- 4. Interpret the discipline to develop personal meaning.
- 5. Know the recommendations of national and state Standards for school content in their discipline.
- 6. Know the major themes and central content of their discipline, as defined for their teaching level, by national Standards.
- 7. Know the major themes and central content of their discipline, as defined for high school

# graduation in Minnesota's Graduation Standards.

Fostering a "deep knowledge" for beginning teachers does not necessarily require additional content courses. Instead, institutions must examine what is learned in content courses and the manner in which content is taught. Programs that produce accomplished beginning teachers will provide experiences that allow them to learn the discipline as mathematicians and scientists do, while making connections within a discipline, across disci-plines, and beyond institutions to the world at large. Faculty who support beginning teachers will examine their role in helping teachers develop the knowledge and skills necessary to teach according to the vision of the *Standards*. Institutions will provide faculty and learners with the kinds of support they need.

### A. WHAT MUST TEACHERS KNOW AND BE ABLE TO DO?

Beginning teachers of mathematics and science will:

#### 1. Understand the fundamentals of

Teachers' comfort with and confidence in their content knowledge influences how and what they teach." (NCTM, 1991)

*(*)

#### their discipline.

- a. Know and use the basic facts, concepts, principles, laws, and theories of the discipline.
- b. Know and use the skills and processes of scientific inquiry or mathematical reasoning to solve problems.
- c. Know and use the tools and techniques of mathematics and science.
- d. Be able to locate sources of information on research in mathematics and science.
- e. Be aware of reasonable and safe practice in a laboratory setting, and understand the consequences of inappropriate practice.

# **2.** Understand the nature of their discipline from a broader perspective.

- Make connections across the sciences, between mathematics and science, and connect mathematics and science with other disciplines and technology.
- b. Know how the content-specific knowledge of mathematics and science is unique from other disciplines.
- c. Know how their discipline relates and applies to the world, and the role of their discipline in social, cultural, and economic development.
- d. Understand the dynamic nature of mathematics and science as a result of new discoveries and new technology.
- e. Know the history and philosophy of their discipline.
- f. Understand the contributions of different cultures toward the growth and development of mathematics and science.

#### 3. Apply the discipline in their own lives.

- a. Use scientific and mathematical knowledge as a basis for decision making when dealing with personal needs and societal issues.
- b. Develop their own processes, concepts, and techniques for solving problems.
- c. Use tools and technology for personal research and learning.

# 4. Interpret the discipline to develop personal meaning.

- a. Communicate effectively in the discipline at different levels of formality and with different audiences.
- b. Know and use analogies, illustrations, and examples in their discipline.
- c. Know what makes the learning of specific concepts difficult or easy.
- d. Know common initial beliefs and alternative theories in their discipline.
- e. Know and use multiple ways of representing concepts.
- 5. Know the recommendations of national and state Standards for school content in their discipline.
- 6. Know the major themes and central content of their discipline, as defined for their teaching level, by national Standards.
- a. Understand the major themes of mathematics: problem solving, reasoning, communication, and mathematical connections.



# Section I - KNOWING MA THEMATICS AND SCIENCE, continued

b. Understand and use number systems and number sense including: mental mathematics, estimation, reasonableness

of results, basic number theory, and the role of algorithms and place values.

- IN MATHEMATICS c. Understand geometry concepts including: shape, space, spatial visualization, geometry from synthetic, coordinate, and transformational perspectives.
- d. Know and use concepts of measurement, especially from the perspective of its historical development.
- e. Understand and use key concepts of statistics and probability.
- f. Understand functions and the use of variables and their use in the communication and growth of mathematical ideas.
- g. Understand algebraic systems and structures.
- h. Understand concepts of calculus.
- i. Use tools and modeling applications of discrete mathematics.
  - IN SCIENCE
- a. Understand the nature of scientific inquiry, and be able to conduct scientific investi-gations appropriate to their level of teaching.
- b. Understand concepts in the physical sciences including: properties of objects and materials; position and motion of objects; light, heat, electricity, and magnetism; properties and changes of properties of matter; motions and forces; transformations of energy; structure of the atom; structure and properties of matter; chemical

reactions; conservation of energy and increase in disorder; and interactions of energy and matter.

- c. Understand concepts in the life sciences including: characteristics of organisms; life cycles of organisms; organisms and environments; structure and function in living systems; reproduction and heredity; regulation and behavior; populations and ecosystems; diversity and adaptations of organisms; the cell; the molecular basis of heredity; biological evolution; the interdependence of organisms; matter, energy, and organization in living systems; and, the nervous system and behavior of organisms.
- d. Understand concepts of earth and space science including: properties of Earth materials; objects in the sky; the structure of the Earth system; Earth's history; Earth in the solar system; energy in the Earth system; geochemical cycles; origin and evolution of Earth system; and, origin and evolution of the universe.
- Understand about science and technology;
  distinguish between natural and human-made
  objects; and have abilities of technological design.
- f. Understand science in personal and social perspectives including: personal and community health; characteristics of changes in and the growth of populations; types of resources; changes in environments; environmental quality; natural and human-induced hazards; risks and benefits; and science and technology in local, national, and global challenges.
- g. Understand science as a human endeavor;
  the nature of science and scientific knowledge; and
  the history of science and historical perspectives.
- h. Understand the unifying concepts and processes of science including: order and organization; evidence, models, and explanation; change, constancy, and

measurement; evolution and equilibrium; and, form and function.

7. Know the major themes and central content of their discipline, as defined for high school graduates in Minnesota's Graduation Standards.

#### IN MATHEMATICS

- a. Analyze patterns and use concepts of algebra to represent mathematical relationships in a variety of ways to solve problems.
- b. Analyze and use discrete structures to represent mathematical relationships in a variety of ways to solve problems.
- c. Apply concepts of shape and space to illustrate and describe the physical world and solve problems.
- d. Apply concepts of chance and data handling to evaluate and solve problems.
- e. Apply mathematical concepts to technological problems and/or the creation of new products.
- f. Apply precision measurement instruments and data handling techniques to technical processes.
- g. Use mathematical problem solving and reasoning to analyze information when conducting an investigation.

#### IN SCIENCE

- a. Understand the interactions and interdepen-dence of components of biological systems.
- b. Understand the concepts, theories, and interactions of chemical structures and the properties of matter.
- c. Understand the interactions and interdepen-dence of components of Earth systems.
- d. Understand the interactions and interdepen-dence of components of physical systems.

- e. Apply decision-making model(s) to issues involving relationships among the individual, the society, the economy, and the environment.
- f. Apply knowledge, skills, and tools of techno-logical systems to extend human capabilities while preserving ecological functions.
- g. Understand the processes by which scientific hypotheses are formulated and tested.
- h. Comprehend and evaluate reports of events or ideas in the context of scientific knowledge.
- i. Understand the interactions between social, economic, technological, and/or environ-mental factors, and the occurrence of scientific advances.
- j. Analyze decisions regarding personal wellness based on scientific understanding of the human body.
- B. WHAT EXPERIENCES SHOULD TEACHERS HAVE TO DEVELOP THEIR KNOWLEDGE AND SKILLS?

Programs which produce beginning teachers of mathematics and science provide experiences for learners to:

#### 1. Learn as mathematicians and scientists do.

- a. Actively investigate scientific and mathematical phenomena.
- Engage in scientific inquiry, mathematical reasoning, and problem solving in laboratory settings with research projects and through examination of research journals.
- c. Investigate issues, events, problems, or topics that are significant to the discipline, important to the community, or of personal interest to learners.
- d. Reflect on the content and how they are learning it.



## Section I - KNOWING MA THEMATICS AND SCIENCE, continued

- e. Engage in significant discourse about mathematics and science.
- f. Use materials such as technology, manipula-tives, scientific and mathematical literature, and media when appropriate to solve problems and broaden the scope of inquiry.
- g. Locate, evaluate, and use secondary sources of data (i.e., government documents, experts, databases) as sources for inquiry.
- h. Participate in classrooms which model reasonable and safe practice, and provide adequate supervision.
- i. Participate in classrooms which model alternative strategies for instruction.
- 2. Make connections within a discipline, across disciplines, and with the world.
- a. Determine the central themes of the discipline.
- b. Determine the inter-relationships among science, mathematics, technology, and society.
- c. Explore examples of the historical development of ideas to understand how information is created.
- d. Uncover the dynamic nature of the discipline including how knowledge is generated and substantiated.
- e. Compare various knowledge frameworks and ways of thinking about the discipline.
- Explore multiple perspectives of mathematics and science including controversies within the disciplines, contributions by both males and

females, and belief systems of other cultures.

- g. Investigate the national and state *Standards* in their discipline.
- C. WHAT ARE THE ROLES AND RESPONSIBILITIES OF F ACULTY IN HELPING TEACHERS DEVELOP THEIR KNOWLEDGE AND SKILLS?

Faculty will:

- 1. Model effective teaching by using a variety of techniques such as authentic assessment, collaborative learning, and writing.
- 2. Model scientific inquiry, mathematical reasoning, and problem solving.
- 3. Encourage and support learners to take risks as they work independently or cooperatively at solving problems.
- 4. Encourage full participation and continuing study of mathematics and science by all learners.
- 5. Encourage reflection both on the nature of the content and on the mechanisms used for learning.
- 6. Include structured opportunities that build on learners' existing knowledge, skills, and beliefs that

help them make sense of new subject matter.

 7. Remain open to current thinking and research on effective teaching in mathematics and science.
 D. WHAT ARE THE CHARACTERISTICS OF INSTITUTIONS THAT PRODUCE ACCOMPLISHED TEACHERS?

Institutions that produce accomplished teachers:

- 1. Possess a vision of mathematics and science teaching and learning aligned with state and national Standards.
- 2. Integrate and coordinate teacher education program components so that understanding can be built over time, reinforced continuously, and practiced in a variety of settings.
- 3. Focus on content and pedagogy simultaneously, e.g., through team teaching by discipline and pedagogy experts, or by enrolling learners simultaneously in content and teaching methods courses.
- 4. Foster collaboration and mutual respect among all stakeholders of the program including beginning teachers, parents/ guardians, teacher practioners, teacher educators, administrators, policymakers, business people, scientists, and mathematicians.
- 5. Assess programs continuously from multiple perspectives using a variety of strategies.

"Mathematics [and science] pedagogy focuses on ways in which teachers help their students come to understand and be able to do and use mathematics [and science]." (NCTM, 1991)

Several essential components of pedagogy serve as lenses through which teachers filter their knowledge of the discipline in order to make effective decisions about teaching. Mastery of pedagogy enables teachers to transform content knowledge into powerful and productive learning experiences that are appropriate for diverse groups of students.

In this section, the knowledge base associated with the mastery of pedagogy required of beginning mathematics and science teachers is organized into six categories. Beginning teachers will:

- 1. Develop a rationale for making decisions about instruction.
- 2. Know and use instructional resources.
- 3. Know and use instructional strategies.
- 4. Know and use strategies to promote discourse and foster a learning community.
- 5. Know and use means to assess student understanding.
- 6. Understand and use pedagogical content knowledge.

Pedagogical content knowledge is the integration of a



# Section II - KNOWING PEDAGOGY IN MATHEMATICS AND SCIENCE

teacher's subject matter knowledge and pedagogical knowledge. It is

the form of knowledge that sets mathematics and science teachers apart from teachers in other disciplines, and from mathematicians or scientists. It is not the quality or quantity of discipline knowledge that is unique, but rather the teacher's ability to represent topics in the discipline in ways that make those topics comprehensible to others.

To help beginning teachers transform their understanding of discipline knowledge and general pedagogy into pedagogical content knowledge requires that programs provide beginning teachers with a variety of experiences. These experiences in content courses, education courses, and schools must be interconnected

to achieve the vision of national and state *Standards*. Professionals who support accomp-lished teachers will examine their roles and responsibilities in this transformation process, while institutions provide faculty and learners with the kinds of support they need.

### A. WHAT MUST TEACHERS KNOW AND BE ABLE TO DO?

Beginning teachers of mathematics and science will:

# 1. Develop a rationale for making decisions about instruction.

a. Know national and state *Standards* and their recommendations for the discipline(s).

- b. Understand and apply current learning theories, especially as they pertain to mathematics and science.
- c. Know and apply the research on effective teaching, especially in mathematics and science.
- d. Be aware of research in mathematics and science and its impact on teaching.
- e. Understand the dynamic nature of mathematics and science as a result of new discoveries and new technology.

#### 2. Know and use instructional resources.

- a. Know and use a variety of professional resources to support content and pedagogy.
- b. Select and evaluate appropriate materials according to the purpose of instruction, the content goals, and the needs of students.
- c. Use technology and other tools appropriately for teaching and learning.

#### 3. Know and use instructional strategies.

a. Know and use a variety of instructional strategies to meet the diverse needs of students.

\*\* Learning involves making connections between new information and existing systems of knowledge; teaching should facilitate making these connections by helping students to relate new knowledge to knowledge they have already developed."

(BROWN & BORKO, 1992)



26 Chanter 2 · Standards

- b. Facilitate higher-level thinking and learning for all students.
- c. Understand and use techniques such as writing in the discipline, concept mapping, and journaling to promote learning.
- d. Elicit and respond to students' initial beliefs and alternative theories about mathematics and science.
- e. Help students build connections between previous experiences and classroom instruction.
- f. Know how to create or modify lessons appropriate to the content.

# 4. Know and use strategies to promote discourse and foster a learning community.

- a. Facilitate cooperative learning and orchestrate classroom discourse.
- b. Use techniques for classroom management that promote cooperation and mutual respect among students and teachers.
- c. Affirm and support full participation and continued study of mathematics and science in ways appropriate for all students.
- d. Develop a positive learning environment that fosters a community of learners in mathematics or science.
- 5. Know and use means to assess student understanding.
- a. Design and evaluate assessment tasks that adequately reflect the content being taught.
- Know multiple techniques for developing, administering, and analyzing assessments that are aligned with instruction.
- c. Make decisions about instruction based on the results of a variety of assessment techniques.

d. Be able to communicate assessment results to students, parents/guardians, and the greater community.

# 6. Understand and use pedagogical content knowledge.

- a. Translate key concepts for students by means of multiple representations.
- b. Know the most powerful analogies, illustrations, examples, and demonstrations in their discipline.
- c. Be able to analyze potential difficulties in the learning of concepts and processes.
- d. Know common student initial beliefs and alternative theories in mathematics and science.

### B. WHAT EXPERIENCES SHOULD TEACHERS HAVE TO DEVELOP THEIR KNOWLEDGE AND SKILLS?

Programs which produce beginning teachers of mathematics and science provide experiences for learners to:

- 1. Develop a rationale statement for teaching mathematics and science.
- 2. Utilize national goals, state, and local mandates as mechanisms to develop a framework of goals for student learning.
- 3. Read and interpret research on teaching mathematics and science.
- 4. Locate, select, and evaluate teaching ideas in professional journals.



## Section II - KNOWING PEDAGOGY, continued

- 5. Observe a variety of exemplary teachers reflecting on the tasks, discourse, environment, and assessments they use.
- 6. Make connections between research on teaching and classroom practice in specific teaching situations.
- 7. Use various techniques such as interviews, journals, and concept maps to understand student thought.
- 8. Select appropriate content and modify curricula to meet the particular interests, knowledge, skills, and experiences of students.
- 9. Critique current instructional materials and resources such as software, textbooks, manipulatives, instructional kits, professional journals, and specialized curricula.
- 10. Develop and evaluate curriculum, instruction, and assessment plans that, when implemented through teaching, embody the spirit of the national Standards in mathematics and science.
- 11. Experience a context of professional practice in which discussions about issues of teaching and learning are closely connected with the everyday work of teaching in schools.
- C. WHAT ARE THE ROLES AND RESPONSIBILITIES OF F ACULTY IN HELPING TEACHERS DEVELOP THEIR KNOWLEDGE AND SKILLS?

Faculty will:

by using a variety of techniques such as authentic assessment, collaborative learning, and writing.

- 2. Encourage and support learners to take risks as they work both independently or cooperatively.
- 3. Utilize exemplary instructional materials and appropriate technology.
- 4. Encourage reflection on the mechanisms used for teaching and learning.
- 5. Acknowledge learners' developmental needs and build on existing knowledge of content, teaching, and learning.

### D. WHAT ARE THE CHARACTERISTICS OF INSTITUTIONS THAT PRODUCE ACCOMPLISHED TEACHERS?

Institutions that produce accomplished teachers:

- 1. Possess a vision of mathematics and science teaching and teacher development aligned with state and national Standards.
- 2. Integrate and coordinate teacher education program components so that a knowledge base can be built over time, reinforced continuously, and put into practice in a variety of settings.
- 3. Focus on content and pedagogy simultaneously, e.g., through team teaching by discipline and pedagogy experts, or by "shadow courses" in which beginning teachers enroll simultaneously in content and teaching methods courses.

1. Illustrate and model effective teaching

- 4. Integrate teacher education coursework with opportunities to assume a variety of experiences with K–12 students from the beginning of the teacher education program.
- 5. Develop collaboration and mutual respect among all stakeholders of the program, including beginning teachers, parents/ guardians, teacher practitioners, teacher educators, administrators, policy-makers, business people, scientists, and mathematicians.
- 6. Foster partnerships between teacher education faculty and discipline specialists for supervision of field experiences.
- 7. Assess programs continuously from multiple perspectives using a variety of strategies.

The focus of national and state *Standards* on what all students of mathematics and science should know and be able to do places a special emphasis on teachers' understanding of students. The Interstate New Teacher Assessment and Support Consortium Standards document best describes this emphasis:

"Teachers work to create learning environments in which students' diverse interests, linguistic, cultural, and socioeconomic backgrounds and special educational needs are respected, and in which the full participation and continued study of mathematics (and science) by ALL students is encouraged."

(INTASC, 1995)

In this section, the knowledge base associated with knowing students as learners is organized into four categories. Beginning teachers will:

- 1. Develop a rationale for making decisions about learners.
- 2. Appreciate students as individuals.
- 3. Recognize the implications of student diversity on learning.
- 4. Foster student learning.

Fostering knowledge of students as learners of mathematics and science may require institutions to change the types of experiences they provide in teacher education programs. Institutions should examine their programs to ensure that experiences with K–12 students begin early and continue throughout the undergraduate teacher education program. These experiences will involve interactions among beginning teachers and university faculty with a diverse range of K–12 students. Faculty involved in all aspects of teacher education will examine their roles and responsibilities in helping beginning teachers learn about students, while institutions provide faculty and learners with the kinds of support they need.

### A. WHAT MUST TEACHERS KNOW AND BE ABLE TO DO?

Beginning teachers of mathematics and science will:

- 1. Develop a rationale for making decisions about learners.
- a. Know and apply current national and state *Standards*-based recommendations for students.


## Section III - KNOWING STUDENTS AS LEARNERS OF MA THEMATICS AND SCIENCE

b. Know and apply the research on student learning, especially the learning of mathematics and science.

### 2. Appreciate students as individuals.

- a. Understand that individuals learn in different ways.
- b. Understand the principles of human development including characteristics, patterns, and variations of normal physical, emotional, intellectual, behavioral, and social development.
- c. Understand that individual students have unique beliefs about the world which influence their interests and expectations.
- d. Understand the impact of teachers' beliefs, attitudes, and actions on students.

## 3. Recognize the implications of student diversity on learning.

- a. Know how students' multiple perspectives, due to cognitive and physical development, may affect learning.
- b. Know how student diversity (linguistic, ethnic, racial, socio-economic, gender) affects learning.
- 4. Foster student learning.
- a. Know and use a variety of instructional strategies.

- b. Know and use a variety of professional resources.
- c. Know how to identify and elicit students' beliefs about the discipline.
- d. Possess a repertoire of pertinent, culturally relevant examples, analogies, and metaphors.
- e. Understand how language acquisition skills are used to teach mathematics or science to students whose primary language is not English.

## B. WHAT EXPERIENCES SHOULD TEACHERS HAVE TO DEVELOP THEIR KNOWLEDGE AND SKILLS?

Programs which produce beginning teachers of

\*Beginning teachers need to understand that the purpose of the school is the education and development of the whole child."

(TTE CONFERENCE, JANUARY 1994)



mathematics and science provide experiences for learners to:

- 1. Participate in experiences with students of various ages and abilities to develop understanding of the differences in individual students.
- 2. Participate in experiences with K–12 students that reflect cultural, ethnic, and socio-economic diversity.
- 3. Use various techniques such as interviews and concept maps to elicit student beliefs about the discipline.
- 4. Listen and talk with students as they construct personal meanings of mathematics and science concepts.
- 5. Observe an individual student over time, and analyze the student's thinking and beliefs about mathematics and science.
- 6. Observe and assist teachers who work with students from different backgrounds and are effective in the teaching of mathematics and science.
- 7. Discuss with effective teachers how they make personal decisions about student learning situations.
- 8. Read and evaluate research on student diversity and effective instructional strategies.
- 9. Discuss mathematics and science education research, particularly as it applies to diverse student populations, with peers, faculty, teacher practioners, and scientists or mathematicians.

- 10. Examine state and national Standards to identify the recommendations they make about students.
- 11. Select and create instructional materials that reflect ethnic and cultural inclusion.
- 12. Identify and practice various strategies that encourage full and active participation of all students in mathematics and science.
- 13. Modify the context of mathematics and science content to better relate it to the lives of students from different backgrounds.
- C. WHAT ARE THE ROLES AND RESPONSIBILITIES OF F ACULTY IN HELPING TEACHERS DEVELOP THEIR KNOWLEDGE AND SKILLS?

Faculty will:

- 1. Design experiences with K–12 students which align with the vision of state and national Standards.
- 2. Select appropriate mentor teachers who can provide valuable learning experiences for beginning teachers.
- 3. Form professional partnerships with school faculty who serve as mentors of beginning teachers.
- 4. Design student teaching to expand upon previous experiences with K–12 students.
- D. WHAT ARE THE CHARACTERISTICS OF INSTITUTIONS THAT PRODUCE ACCOMPLISHED TEACHERS?



## Section III - KNOWING STUDENTS AS LEARNERS, continued

Institutions that produce accomplished teachers:

- Possess a vision of mathematics and science teaching and learning aligned with state and national Standards.
- 2. Have diverse faculty and student populations in undergraduate teacher education programs.
- 3. Foster substantive, professional discussions among discipline-based and professional education faculty.
- Foster substantive, professional discussion among university and K-12 faculty.
- Initiate beginning teachers into a professional community of educators (i.e., MSTA, MCTM, MFT, MEA).
- 6. Provide school experiences that begin early (no later than the sophomore year) and continue throughout the undergraduate teacher education program.
- 7. Connect school experiences with the development of the knowledge base about students from different backgrounds.

The knowledge and skills needed to create an emotionally safe and supportive environment were presented in Sections II and III. Another aspect of establishing an environment for learning mathematics and science includes

the physical arrangement of the classroom.

Teachers should organize the physical environment of the classroom to ensure that students have the space, resources, and time they need to do science and mathematics. Yet learning environments also extend beyond classroom walls. For teachers to be effective implementors of the *Standards*, they must also be able to use community and global resources for instruction.

Establishing safe and supportive learning environments for students requires that teachers also feel safe and supported in their teaching environment. To do this, they need

to understand the school and community environments in which they teach. This means being able to use strategies for learning about the culture of the local school and community. Teachers need to understand that effective teaching requires the support of individuals beyond the classroom, including adminis-trators, parents/guardians, school staff, and other faculty.

In this section, the knowledge base needed by beginning teachers of mathematics and science to establish an environment for learning is organized into four categories. Beginning teachers will:

### 1. Develop a rationale for making decisions about



learning environments.

- 2. Establish a physical environment for learning.
- 3. Use community and global resources for instruction.
- 4. Understand the school/community environment for teaching.

Fostering the knowledge and skills teachers need to establish a safe and supportive learning environment requires that institutions provide experiences for beginning teachers that allow them to acquire greater exposure

to schools and other community settings. Institutions must also form partnerships with business and industry to provide the resources necessary to make institutional classrooms and laboratories model sites for safe, accessible mathematics and science teaching.

# A. WHAT MUST TEACHERS KNOW AND BE ABLE TO DO?

Beginning teachers of mathematics and science will:

- 1. Develop a rationale for making decisions about learning environments.
- Know national and state *Standards* recommendations for learning environments in their discipline(s).
- b. Know the positions of state and national professional organizations on safe learning environments.



## Section IV - ESTABLISHING AN ENVIRONMENT FOR LEARNING MA THEMATICS AND SCIENCE

- c. Know state and national legislation regarding a safe workplace.
- 2. Establish a physical environment for learning.
- a. Know and apply state and national guidelines for use of chemicals, organisms, and equipment in laboratory settings.
- b. Know guidelines regarding personal liability in a laboratory setting.
- Know the liabilities and responsibilities of the administration and school board for a safe laboratory.
- d. Know characteristics of laboratory settings that provide students with appropriate access to the tools and materials needed for hands-on mathematics and science.
- e. Know features of classrooms organized to meet the needs of students with disabilities.
- f. Know safety equipment that must be available in classroom or laboratory settings, and ensure appropriate use.
- g. Know procedures for correcting safety deficiencies.
- h. Acquire and maintain materials for classroom use such as educational software, instructional kits, and videos.
- i. Know of and use alternative settings for instruction.

- j. Know of and use professional organizations and publications for decisions about equipment and supplies.
- 3. Use community and global resources for instruction.
- a. Know of important real-world events, problems, and experiences to set contexts and provide learning.
- b. Know of traditional and non-traditional sites such as museums, nature centers, and local industry that can be used for formal and informal instruction.
- c. Know and use local resources such as speakers and community organizations.

# 4. Understand the school/community environment for teaching.

- a. Know and use strategies to determine what a school expects of teachers.
- b. Know and use strategies to determine the demographics (socio-economic, safety, crime, etc.) of the school community.
- c. Know how others (administrators, the community, parents/guardians, other teachers, staff, and professional organizations) can impact their

**\*\*We** believe that professional teachers assume roles that extend beyond the classroom and include responsibilities for connecting to parents and other professionals, developing the school as a learning organization, and using community resources to foster the education and welfare of students." (INTASC, 1992)



teaching.

- d. Know where or how to find the philosophy, mission, and goals of a district.
- e. Understand how district goals are determined and assessed.
- f. Know how a school system operates.
- g. Know how to ask questions and make observations about the political and cultural climates, including community beliefs and attitudes within the school.
- h. Understand the central role of state and local politics in funding and policy related to mathematics and science education.

## B. WHAT EXPERIENCES SHOULD TEACHERS HAVE TO DEVELOP THEIR KNOWLEDGE AND SKILLS?

Programs which produce beginning teachers of mathematics and science provide experiences for learners to:

- 1. Review national and state **Standards** for recommendations regarding a safe and supportive learning environment.
- 2. Develop model laboratory settings that provide students with appropriate access to the tools and materials needed for hands-on mathematics and science.
- 3. Examine and evaluate systems for the safe organization and storage of materials needed to do

hands-on mathematics and science.

- 4. Examine various designs for effective classroom and laboratories that meet the needs of all students.
- 5. Examine catalogs and other resources for materials to teach mathematics and science.
- 6. Participate in classrooms which model reasonable care and safety, and provide adequate supervision and appropriate use of laboratory safety equipment.
- 7. Develop a file of community resources for a typical community.
- 8. Investigate the political and social culture of a school within a community, and its impact on students and learning.
- C. WHAT ARE THE ROLES AND RESPONSIBILITIES OF F ACULTY IN HELPING TEACHERS DEVELOP THEIR KNOWLEDGE AND SKILLS?

Faculty will:

- 1. Develop an understanding of current school environments.
- 2. Develop connections with schools.



## Section IV - ESTABLISHING AN ENVIRONMENT, continued

- 3. Teach in settings that are safe, accessible, and appropriate for instruction.
- 4. Know the resources of the community and utilize them, as appropriate, for instruction.

## D. WHAT ARE THE CHARACTERISTICS OF INSTITUTIONS THAT PRODUCE ACCOMPLISHED TEACHERS?

Institutions that produce accomplished teachers:

- 1. Possess a vision of mathematics and science teaching and learning aligned with state and national Standards.
- 2. Center programs in school or other settings that represent real teaching situations.
- 3. Organize teacher education programs around teachers' experiences, case studies, and artifacts of schooling.
- 4. Teacher education courses should be taught by faculty with strong K–12 connections.
- 5. Form collaborations with schools, institutions, and

businesses to prepare mathematics and science teachers who can use all the resources of the community.

Professional development is a "life-long" learning process that begins with the first course in a beginning teacher's undergraduate program. As described in this quote, professional development

"... is a dynamic process extending from initial preparation over the course of an entire career. Professional teachers are responsible for planning and pursuing their on-going learning, for reflection with colleagues on their practice, and for contributing to the profession's knowledge base." (INTASC, 1992)

As an introduction to professional development, teachers need on-going opportunities to observe and talk with master practitioners on a regular basis. They should be introduced to the skills, knowledge, and dispositions they need to engage in life-long learning. This includes the previously defined knowledge base related to content, pedagogy, students, and learning environments along with other professional development topics presented in this section.

In this section, the knowledge base needed by beginning teachers of mathematics and science to begin the process of "life-long learning" is organized into five categories. They are:

- 1. Develop a rationale for making decisions about professional development.
- 2. Act as knowledgeable and effective consumers of research as it relates to their teaching and to their students.
- 3. Understand the nature of professional development.
- 4. Act as reflective practitioners.



# 5. Identify and use resources for professional development.

Fostering professional development requires that institutions help each beginning teacher develop as a reflective practitioner, and use this reflection to impact practice. As teachers grow and develop as professionals, they need experiences that begin in teacher preparation institutions, and they need to move into schools that serve as models for effective practice. Faculty and others who support the professional development of teachers must examine their roles and responsibilities in this process. In order to provide a seamless transition between preservice and inservice experiences, institutions must have a vision of mathematics and science teaching aligned with state and national Standards, and provide faculty and learners with the kinds of support they need to be effective.

When reading this section, it is important to note that the professional development of beginning teachers must be shared with other responsible stakeholders, including teacher practioners, K–12 schools, the community,

and professional organizations.

## A. WHAT MUST TEACHERS KNOW AND BE ABLE TO DO?

Beginning teachers of mathematics and science will:

- 1. Develop a rationale for making decisions about professional development.
- a. Know the recommendations of national and state *Standards.*
- b. Know the recommendations of state and national professional organizations.
- c. Know related research.
- 2. Act as knowledgeable and effective consumers of research as it relates to their teaching and students.
- 3. Understand the nature of professional



Chanter 2 3

## Section V - DEVELOPING AS A TEACHER IN MA THEMATICS AND SCIENCE

### development.

- a. Understand that learning mathematics and science is a career-long process.
- b. Know that becoming an effective teacher of mathematics and science is a continuous process that stretches to retirement.
- c. Understand the role of a learning community in professional growth and development.
- d. Understand the teacher's role in changing school culture and practice.
- e. Understand the benefits and opportunities of memberships in mathematics and science teachers' organizations.

### 4. Act as reflective practitioners.

- a. Use tools for reflection including journals, audio tapes, videotapes, and portfolios.
- b. Conduct investigations into the effectiveness of their teaching.

# 5. Identify and use resources for professional development.

- a. Identify and evaluate the usefulness of popular and professional literature in mathematics, science, and education.
- b. Develop and utilize professional networks of colleagues.
- c. Use communication technology to access and assure continuing professional development.
- d. Locate opportunities (workshops, conferences, and courses) for professional development in mathematics, science, and education.

## B. WHAT EXPERIENCES SHOULD TEACHERS HAVE TO DEVELOP THEIR KNOWLEDGE AND SKILLS?

Programs which produce beginning teachers of mathematics and science provide experiences for learners to:

- 1. Review national and state Standards for recommendations regarding professional growth and development.
- 2. Reflect on learning and teaching individually and with colleagues at various stages of professional development.
- 3. Begin and sustain interaction with master practitioners such as mentors, teacher advisors, coaches, lead teachers, and resource teachers.
- 4. Explore alternative instructional strategies, and receive feedback about their teaching.
- 5. Read, react to, and discuss professional publications in mathematics and science education.
- 6. Participate in electronic networks focused on mathematics and science teaching.

**General Schultz and Schultz a** 



- 7. Participate in state or regional meetings of professional mathematics and science organizations.
- 8. Become involved with professional organizations in mathematics and science education.
- 9. Participate in personal reflections on the teacher education program at their institution.
- C. WHAT ARE THE ROLES AND RESPONSIBILITIES OF F ACULTY IN HELPING TEACHERS DEVELOP THEIR KNOWLEDGE AND SKILLS?

Faculty will:

- 1. Serve as strong models for professional growth and development. This includes faculty from the institution and at field sites.
- 2. Participate in scholarly and professional activities related to effective teaching.
- 3. Model service to professional organizations. This includes both discipline-based and professional teaching faculty.
- D. WHAT ARE THE CHARACTERISTICS OF INSTITUTIONS THAT PRODUCE ACCOMPLISHED TEACHERS?

Institutions that produce accomplished teachers:

- 1. Possess a vision of mathematics and science teaching and learning aligned with state and national Standards.
- 2. Consider professional development as a key component of teacher education.
- 3. Understand the professional develop-ment of beginning teachers as the shared responsibility of the entire education community, including the institution, teacher practioners, K–12 schools, state education organizations, and professional organizations.
- 4. Provide professional development opportunities for teachers that clearly and appropriately connect to teachers' work in the context of the school.
- 5. Ensure alignment between undergraduates and K-12 experiences for beginning teachers by working in partnership with local schools.



## Section V - DEVELOPING AS A TEACHER, continued

- 6. Establish and support a beginning teacher network for recent graduates.
- 7. Make institutional resources available to teachers after graduation.
- 8. Value and reward faculty participation in professional development related to effective teaching.
- 9. Serve as models for change by continuously evaluating and improving teacher preparation programs.
- 10. Solicit feedback from individuals who were part of the institution's teacher education program.

is not optional in such a profession [mathematics and science teaching]; it is rather an integrated part of the practice of one's art."

MARY PEARLMAN

**"The notion of** being a 'life-long learner'





[BLANK PAGE]



Chapter 3

## Recommendations

for Transforming the Education of Teachers of Mathematics and Science

### INTRODUCTION

The next decade will be a time of excitement and challenge for Minnesota teacher preparation institutions. New discipline knowledge, new technology, and new understanding of teaching and learning will provide institutions with the information they need to address the vision created by state and national *Standards* for teacher preparation.

The previous chapter described a framework for achieving Minnesota's vision for accomplished beginning mathematics and science teachers. Standards for teacher experiences were provided to develop the knowledge and skills teachers need to facilitate all K–12 students in the attainment of scientific and mathematical literacy. Descriptions of faculty roles and responsibilities needed to support beginning teachers were also presented, along with characteristics of institutions that produce accomplished beginners.

The next steps require that institutions begin a process of self-assessment. First, teams of teacher education stakeholders at each institution must closely examine the Minnesota vision, underlying state and national *Standards*, and necessary conditions in Minnesota for teacher licensure. Then, institutions should review their own unique needs, realities, and goals for teacher education. Finally, institutions should examine the align-ment of their own teacher education programs with the vision presented here, and determine where change is necessary and possible in light of the Minnesota rules for teacher licensure.

The recommendations in this chapter relate to the processes of trans-formation, implementation, and continuous renewal of a quality teacher education program. The transformation process is complex. It requires systemic change both within the institution and beyond. Recent research related to systemic change stresses the importance of a shared leadership model in which the unique contributions and efforts of every stakeholder are utilized in the process of reform. The pages which follow define the unique roles and responsibilities of faculty, institutions, Minnesotans, government, business and industry, professional organizations, and K–12 schools/districts.

## A. MODEL DEVELOPMENT OF PROFESSIONAL NETWORKS.

Faculty will:

1. Develop sustained connections with

for the outcome, but you are responsible for the effort."

MARY JO COPELAND

K Never doubt that a small group of thoughtful, committed individuals can change the world: indeed it is the only thing that ever has."

MARGARET MEAD

## Section I - WHAT SHOULD F ACULTY DO TO EDUCATE ACCOMPLISHED TEACHERS?

### K-12 schools and school personnel.

- a. Identify mentor teachers who can provide valuable experiences for beginning teachers.
- b. Form professional partnerships with K–12 faculty who serve as mentors of beginning teachers.
- c. Develop an understanding of the current school culture.
- d. Invite school faculty into discipline and professional education departments to serve as resources for teaching, and to help K–12 teachers develop a better understanding of institutional culture.
- e. Develop professional connections with school staff, teachers, and administrators at field experience sites.
- 2. Develop sustained connections with professional teaching organizations.
- 3. Develop sustained professional connections with faculty in other disciplines.

## B. MODEL PROFESSIONAL GROWTH AND DEVELOPMENT.

Faculty will:

- 1. Understand and apply research on teaching and learning.
- 2. Participate in scholarly activity.

- a. Conduct research related to their discipline.
- b. Use action research and other reflective techniques to continually improve their teaching.
- 3. Maintain an active professional development plan.
- a. Continuously update their knowledge and skills.
- b. Participate in professional activities related to effective teaching.
- 4. Contribute to professional (discipline-based and educational) organizations and agencies.
- 5. Serve as mentors for beginning teachers.

## C. MODEL EXEMPLARY TEACHING.

Faculty will:

- 1. Use active teaching strategies.
- a. Utilize strategies for engaging mathematics and science students in active learning.
- b. Utilize exemplary instructional materials and appropriate technology.
- 2. Utilize authentic assessment aligned with instruction.



**44** Chapter 3 · Recommendations

- 3. Acknowledge and accommodate learners' developmental needs.
- 4. Help learners incorporate new ideas into their existing framework.
- 5. Encourage reflection on the nature of the content, and on the mechanisms used for learning.
- 6. Provide a supportive atmosphere.
- a. Encourage full participation and continuing study of mathematics and science by all learners.
- b. Encourage and support learners to take intellectual risks as they work independently or cooperatively.
- c. Provide settings that are safe, accessible, and appropriate for learning.

## A. MAKE TEACHER EDUCATION A PRIORITY OF THE INSTITUTION.

Institutions will:

- 1. View teacher education as a major responsibility to society.
- 2. Assign highly-qualified faculty to teacher education programs.

## B. ALLOCATE RESOURCES IN A DELIBERATE WAY TO SUPPORT



JOAN L. CURCIO



Chanter 3 4

## Section II - WHAT SHOULD INSTITUTIONS DO TO EDUCATE ACCOMPLISHED

# QUALITY IN TEACHER EDUCA TION PROGRAMS.

Institutions will:

- 1. Support faculty who model a professional approach to teaching.
- a. Sponsor seminars on such topics as teaching, curriculum, assessment, and technology both within and between departments.
- b. Establish department and university support for curricular innovation and quality teaching through the faculty reward system (e.g., salary, promotion, tenure, and load).
- 2. Support faculty involvement in the schools.
- a. Provide resources that allow faculty to spend time in the schools.
- b. Provide professional development opportunities for faculty that clearly and appropriately connect to teachers' work in the context of the school.
- c. Reward school involvement by faculty as valuable contributions to the life of the institution.
- 3. Support school involvement in the university.
- a. Use K-12 teachers to team teach university courses.
- b. Establish K–16 collaboratives.

- c. Use K–12 teachers and administrators as advisors for university programs and practice.
- 4. Provide specialized classrooms and laboratories; state-of-the-art curricular and demonstration materials; and calculators and computer technology at least comparable to those used in the best elementary and secondary schools.
- 5. Hire faculty and recruit student populations in undergraduate teacher education programs who reflect diversity.

### C. ENSURE QUALITY CANDIDATES.

Institutions will:

- 1. Recruit capable students.
- 2. Encourage talented mathematics and science students to consider teaching.
- 3. Maintain high standards for admission to and completion of teacher education programs.
- 4. Assure that teacher education students develop general literacy and thinking abilities.

# D. DEVELOP A STRONG AND INTEGRATED PROGRAM.

Institutions will:



- 1. Include a quality general education program, an in-depth subject matter preparation, and both general and content-specific preparation in teaching methodology.
- 2. Develop strong partnerships with other higher education institutions.
- 3. Integrate and coordinate teacher education program components so that understanding can be built over time, reinforced continuously, and practiced in a variety of settings.
- 4. Integrate coursework in teacher education with opportunities to assume various classroom responsibilities throughout the teacher education program.

### 5. Provide realistic teaching experiences.

- a. Organize teacher education programs around teachers' experiences, case studies, and artifacts of schooling.
- b. Use community resources to complement instruction.
- 6. Provide opportunities for peer support during student teaching.
- 7. Provide continuing support to recent graduates.
- a. Establish and support a beginning teacher network.
- b. Make the resources of the institution available to teachers after graduation.
- E. INVOLVE ALL STAKEHOLDERS.

Institutions will:

# 1. Foster discourse among individuals within the institution and beyond.

- a. Encourage discourse among faculty in mathematics, science, and education; administrators; and teacher practitioners involved in the education of beginning teachers.
- b. Form professional communities for discussion among faculty from discipline and education departments, and from school sites.
- 2. Facilitate beginning teachers initial involvement into a community of professional practice.

## 3. Develop collaboration and mutual respect among all stakeholders of the program.

 a. Foster partnerships among beginning teachers, teacher practitioners, teacher educators, mathematicians and scientists, administrators, policy-makers, and business people for



Chanter 3

## Section II - WHAT SHOULD INSTITUTIONS DO, continued

development of quality programs.

- b. Foster partnerships between teacher education faculty, discipline specialists, and teacher practitioners for supervision of field experiences.
- c. Foster partnerships between teacher education faculty, discipline specialists, and teacher practitioners for planning and teaching courses.
- F. STRESS ACCOUNTABILITY AT ALL LEVELS.

Institutions will:

- 1. Assess programs continuously from multiple perspectives using a variety of strategies.
- 2. Make all departments accountable for their contributions to teacher education.
- 3. Develop equitable methods for evaluating faculty teaching and professional development.
- 4. Define vigorous learning expectations and exit requirements for all learners.
  A. MINNESOTANS SHOULD:
- 1. Develop and model habits for life-long learning.
- 2. Value education and quality teaching.
- 3. Encourage talented individuals to become mathematics and science teachers.
- 4. Recognize the contribution of quality teacher education programs to the quality

of life in Minnesota.

- 5. Be aware of the existence and importance of state and national Standards.
- 6. Recognize teachers as professionals.
- 7. Make a commitment to adequately fund teacher education.

### **B. GOVERNMENT SHOULD:**

- 1. Recognize that the education of mathematics and science teachers is an investment in Minnesota's future.
- 2. Recognize that the preparation of



ADAPTATION OF AFRICAN PROVERB



**48** Chapter 3 · Recommendations

#### Section III - HOW SHOULD SOCIETY **PROVIDE SUPPORT FOR** THE EDUCATION OF ACCOMPLISHED TEACHERS?

quality mathematics and science teachers is a complex process that cannot be accomplished simply by mandate.

- 3. Develop, fund, and enforce teacher licensure and evaluation guidelines consistent with national Standards.
- 4. Make a commitment to develop and fund a coherent vision of quality teacher education.
- 5. Provide significant time and resources to achieve the vision of quality teacher education.
- 6. Understand that educational reform must be systemic, and cannot be accomplished simply by changing K–12 teachers and schools.
- 7. Work with university leaders to support effective teacher education programs.
- C. BUSINESS AND INDUSTRY SHOULD:

- 1. Participate in state and local collabor-ations that are working to improve the education of mathematics and science teachers.
- 2. Communicate the need for mathematically and scientifically literate workers to K-16 educators, administrators, curriculum developers, and faculty.
- 3. Provide internships for beginning teachers to acquaint them with industry resources for teaching.
- 4. Support efforts to secure funds for teacher education.

## D. PROFESSIONAL ORGANIZATIONS SHOULD:

- 1. Form partnerships among K-12 and higher education professional organizations (discipline-based and teaching organizations).
- 2. Sponsor forums on mathematics and science teacher education that draw stakeholders from diverse constituencies.
- 3. Maintain on-going dialogue about the needs of beginning mathematics and science teachers.





## Section III - HOW SHOULD SOCIETY PROVIDE SUPPORT, continued

MARY JEAN LE TENDRE

- 4. Recognize exemplary beginning teachers and teacher education programs in mathematics and science.
- 5. Provide opportunities and encouragement for beginning mathematics and science teachers to participate in conferences and other activities.

## E. K-12 SCHOOLS/DISTRICTS SHOULD:

- 1. Support beginning teachers through mentoring and appropriate assignments.
- 2. Encourage and support professional growth and development.
- 3. Establish and support professional networks.
- 4. Establish partnerships with teacher education institutions for the continuing professional growth and development of all teachers.

**\*\*A**merica's future walks through the doors of our schools each day."





Chapter 4

## Transforming

**Policy to Practice** 

## THE TIME FOR CHANGE

**"E**fforts to bring about lasting change must proceed steadily for many years, on many levels simultaneously, with the broad involvement of all of the constituencies at each stage."

(NRC, 1989)

There has never been a better time in Minnesota to transform teacher education. The *Minnesota Graduation Standards* and new K–12 national standards in mathematics and science have piqued public interest in education. New recommendations are currently being made for initial teacher licensure and continuing professional development at both the state and national levels. Emerging research findings on teacher knowledge and teacher education provide direction

for reform and new educational technologies provide the tools. In addition, many Minnesota institutions

of higher education are now beginning program and curriculum reform as they undergo the change from quarter to semester systems. The shared vision described by this document provides a focus for the transformation of teacher education.

### **Challenges to Meet**

The process of transformation has already begun in Minnesota. Kindled by the process that resulted in this document, widespread dialogues are now occurring across the state between all stake-holders involved in teacher education. But these dialogues are only a first step in the process. Numerous challenges still remain. Many individuals involved in teacher education have strongly held beliefs about how teachers should be prepared. They teach according to these beliefs, they know they are doing a good job, and they see no reason to change. For them, trans-forming teacher education means changing someone else's program.

Existing political frameworks also serve as barriers at both the state and national level. Political decisions are frequently made by committees more concerned about consensus than reform. As a result, the decisions made often do not reflect the intent of state or national reform efforts. Cost and time may be two of the biggest barriers to reform. Both are precious commodities and reform requires a significant commitment of both, often from individuals who have little of either to spare.

Teacher education institutions across the state are at different points in the reform process. Even within a single institution great diversity exists among individuals and departments in their understanding of current recommendations for teacher education and in their willingness to change. The strengths and skills of the various

stakeholders involved in reform also result in variations of purpose and process. Because of these variations,

the process of reform appears to be disjointed. An Organizer for Reform

The pathway of reform may seem less disjointed if one examines the process through an organizer for reform suggested by Rodger Bybee, Executive Director of the Center for Science, Mathematics, and Engineering Education at the National Research Council. He has identified four stages of the reform process in science education. These four stages, which are equally useful in discussing reform in teacher education, are: clarifying the *purposes* of teacher education, establishing *policies* for different aspects of teacher education, developing *programs* for teacher education, and changing *practices* in teacher education classrooms (Bybee, 1995). Applying this model to the Minnesota Transforming Teacher Education process reveals that the first two stages have been achieved.

### PURPOSE

The vision of the Transforming Teacher Education Committee defines a *purpose* for mathematics and science teacher education in Minnesota. It states that:

Beginning mathematics and science teachers will be prepared to teach according to the vision of present and future national standards, and they will be prepared to continue learning new content and new ways of teaching throughout their professional lives.

### POLICY

This document, *Transforming Teacher Education: A Minnesota Framework for Mathematics and Science*, provides *policy* for state efforts to improve teacher education,

in Minnesota. It directs the state's attention toward a common goal for teacher education and provides guidance for institutions as they examine existing programs and make choices for restructuring. It outlines the knowledge, skills, and experiences beginning teachers need in order to teach according to the vision of state and national standards.

### PROGRAM AND PRACTICE

It is the last two stages of Bybee's model that remain to be achieved in Minnesota. Programs are the courses, lessons, tasks, textbooks, experiences, and assessments that are needed to implement stated policy. These will require a great deal of work by many groups and individuals. Furthermore, each component of the programs developed will have to meet the needs of individual institutions. Finally, practice refers to the individual actions of teacher educators in their own classrooms. Achievement of this component will be the most difficult and time consuming as more people are engaged in the process and as more resources are needed. A variety of individuals will need to dedicate long-term personal commitment to the same idea.

### Why Policy is Not Enough

Transforming the policy presented by this document to programs and practice requires an understanding of the reform effort, a commitment to recommendations, and changes in the behavior and beliefs of individual educators. This framework can be a powerful tool to begin the process, but a framework alone is not sufficient to create standardsbased systemic reform in teacher education.

Contemporary educational thought provides several reasons why policy alone is not enough to initiate and



sustain educational reform (Fullan: 1993). First, the document may be misunderstood due to the mindset of the reader. Each reader brings a unique perspective to the document, a perspective developed from the experiences he/she has had with teacher education. These multiple perspectives mean that individuals will interpret the intent of framework statements differently. Depending on the extent of their experiences with new state and national standards, readers may even believe that their current practice is already in alignment with the vision presented for teacher education.

A second reason that policy documents are not sufficient for reform relates to the complexity of the process of reforming teacher education. The changes suggested by any framework represent a complex set of variables involving interactions between faculty from various departments, their curricula, and students from diverse backgrounds. Add to this a need for multiple learning environments including university classrooms, laboratories, K–12 schools and community settings, and the task of reform appears to be overwhelming. The complexity will increase even more as new teacher education programs are developed

and evaluated. Dealing with problems of this complexity is not a task commonly undertaken by most teacher educators.

The third reason has to do with the inherent limitations of any framework. While this framework does provide an extensive description of what beginning teachers should know, experience, and be able to do, it does not describe a process for translating the framework into practice. Nor is it designed to serve as a prescribed curriculum, a set of assessment standards or state licensure requirements. Each of these items must still be developed by teams of teacher educators who will translate the framework into actual programs, courses, curricula, and assessments to meet student needs at individual institutions.

### A Call to Action

Rodger Bybee (1993) suggests four indispens-able requirements for standards-based reform that can be applied to the reform of teacher education in Minnesota. First, we must point all of our efforts in one direction. This framework is an attempt to provide that direction. Second, we must support the standards presented by the framework and use them as a central theme of Minnesota's efforts to reform curriculum, assessment, and instruction in teacher preparation institutions. Third, each

of us must become a spokesperson for the document and for reform in teacher education.

We must explain, not just to our colleagues, but to all Minnesotans that teachers are the

key to enhanced scientific and mathematical literacy for all students. We must tell them why enhanced mathematical and scientific literacy is critical to the



Chanter / 5

well-being of our nation, and that improved teacher education, as described by this framework, is a critical first step in the reform process. Finally, we must all provide support to those who are involved in the process of implementing the changes associated with the framework.

### Conclusion

The transformation of mathematics and science teacher education in Minnesota is an elusive but achievable goal. It will appear to be an insurmountable task only if we fail to work together to accomplish change. Each stake-holder must assume that he/she is a decision maker in the reform process. Each must understand his/her own responsibility in the process, and recognize the long-term commitment that the process requires. When individuals work together as teams, departments, and institutions, those tasks that formerly seemed insurmountable will become

merely challenges to be met and overcome. If each of us accepts our role in

the process, we will be successful.

**\*\*The secret of getting ahead** is getting started."

SALLY BERGER



## References

American Association for the Advancement of Science. (1989). Project 2061, *Science for All Americans.* Washington, DC: Author.

American Association for the Advancement of Science. (1993). *Benchmarks for Scientific Literacy.* Washington, DC: Author.

American Association for the Advancement of Science. (1995). Blueprint for Teacher Education (Working Draft). Washington, DC: Author.

Bybee, R. W. (1993). *Reforming Science Education: Social Perspectives and Personal Reflections*. New York: Teachers College Press, Columbia University.

Bybee, R. W. (1995). Achieving Scientific Literacy. The Science Teacher, 62(7), 28-33.

Fullan, M. (1993). Change Forces: Probing the Depths of Educational Reform. New York: The Falmer Press.

Harms, N., and R. Yager. (1981). What Research Says to the Science Teacher, Vol. III. Washington, DC: National Science Teachers Association.

Hart, L. E., Smith, D. W., Grynkewich, L. C., Primm, S., and Mizelle, N. B. with Jackson, D. F. and Mahaffrey, M. L. (1994). *Principles of Educating Teachers in Mathematics and Science (POETS)*. Athens, GA: Georgia Initiative in Mathematics and Science (GIMS).

Interstate New Teacher Assessment and Support Consortium. (1992). *Model Standards for Beginning Teacher Licensing and Development: A Resource for State Dialogue (Working Draft).* Washington, DC: Council of Chief State School Officers.

 Interstate New Teacher Assessment and Support Consortium. (1995). *Model Standards in Mathematics for Beginning Teacher Licensing and Development: A Resource for State Dialogue (Working Draft).* Washington, DC: Council of Chief State School Officers.

Leitzel, J. R. C., ed. The Mathematical Association of America. (1991). A Call for Change: Recommendations for the Mathematical Preparation of Teachers of Mathematics. Washington, DC: Mathematical Association of America.

Mathematical Association of America. (1991). A Call for Change: Recommendations for the

Mathematical Preparation of Teachers of Mathematics. Washington, DC: Author.

- Minnesota Department of Children, Families and Learning. (June 12, 1995). *Graduation Standards* (*Draft ).* St. Paul, MN: Author.
- The National Commission on Excellence in Education. (1983). *A Nation at Risk: The Imperative for Educational Reform.* Washington, DC: U.S. Government Printing Office.
- National Board of Professional Teaching Standards, Adolescence and Young Adulthood/Mathematics Standards Committee. (July 1993). *Draft Report on Standards for National Board Certification*. Washington, DC: National Board for Professional Teaching Standards.
- National Board of Professional Teaching Standards, Adolescence and Young Adulthood/Science Standards Committee. (July 1994). *Draft Report on Standards for National Board Certification*. Washington, DC: National Board for Professional Teaching Standards.
- National Council of Teachers of Mathematics. (1980). *An Agenda for Action: Recommendations for School Mathematics of the 1980s.* Reston, VA: Author.
- National Council of Teachers of Mathematics. (1989). *Curriculum and Evaluation Standards for School Mathematics.* Reston, VA: Author.
- National Council of Teachers of Mathematics. (1991). *Professional Standards for Teaching Mathematics*. Reston, VA: Author.
- National Council of Teachers of Mathematics. (1995). *Assessment Standards for School Mathematics*. Reston, VA: Author.
- National Research Council. (1989). Everybody Counts: A Report to the Nation on the Future of Mathematics Education. Washington, DC: National Academy Press.
- National Research Council. (1994). *National Science Education Standards (Draft)*. Washington, DC: National Academy Press.
- National Science Teachers of America. (1992). *Standards for Science Teacher Preparation.* Washington, DC: Author.

National Science Teachers Association. (1989). Essential Changes in Secondary School Science: Scope,

Sequence and Coordination of Secondary School Science. Washington, DC: Author.

Richardson, Virginia. (1994). "Standards and Assessments: What is Their Educative Potential?" in *Setting Standards and Educating Teachers: A National Conversation.* Washington, DC: American Association of Colleges for Teacher Education.

Senge, P. (1990). The Fifth Discipline. New York: Doubleday/Currency.

Silberman, C. (1970). Crisis in the Classroom. New York: Random House.

Tyson, H. (1994). *Who Will Teach the Children? Progress and Resistance in Teacher Education.* San Francisco, CA: Jossey-Bass.

### American Association for the Advancement of Science

### Blueprint for Teacher Education (Working Draft). 1995.

For further information contact: American Association for the Advancement of Science 1333 H Street, NW [BLANK PAGE]



## Appendices

## APPENDIX A: NA TIONAL STANDARDS FOR MA THEMATICS AND SCIENCE TEACHING AND TEACHER EDUCA TION

Washington, DC 20005 (202) 326-6400

### Interstate New Teacher Assessment and Support Consortium

Model Standards for Beginning Teacher Licensing and Development: A Resource for State Dialogue (Working Draft). 1992.

Model Standards in Mathematics for Beginning Teacher Licensing and Development: A Resource for State Dialogue (Working Draft). 1995.

For further information contact:

Council of Chief State School Officers 1 Massachusetts Avenue, NW Washington, DC 20001 (202) 408–5505

### Mathematical Association of America

A Call for Change: Recommendations for the Mathematical Preparation of Teachers of Mathematics. 1991.

For further information contact:

Mathematical Association of America 1529 Eighteenth Street, NW Washington, DC 20036

### **National Board of Professional Teaching Standards**

### Draft Report on Standards for National Board Certification. 1994.

For further information contact:

National Board of Professional Teaching Standards 300 River Place, Suite 3600 Detroit, MI 48207

### (800) 228-3224

### **National Council of Teachers of Mathematics**

### Professional Standards for Teaching Mathematics. 1991.

For further information contact: National Council of Teachers of Mathematics 1906 Association Drive Reston, VA 22091-1593 (703) 620–9840

### **National Research Council**

### National Science Education Standards. 1996.

For further information contact: National Research Council 2101 Constitution Avenue, NW Washington, DC 20418 (202) 334–1399

### National Science Teachers Association.

Revitalizing Teacher Preparation in Science: An Agenda for Action. 1994.

#### NSTA Standards for Science Teacher Preparation. 1992.

For further information contact: National Science Teachers Association 1840 Wilson Boulevard Arlington, VA 22201 (703) 243–7100

### American Association for the Advancement of Science

Project 2061: Science for All Americans. 1989.

Benchmarks for Scientific Literacy. 1993.

For further information contact:

American Association for the Advancement of Science 1333 H Street, NW

## APPENDIX B: NA TIONAL STANDARDS FOR CURRICULUM, INSTRUCTION, AND ASSESSMENT IN MA THEMATICS AND SCIENCE

Washington, DC 20005 (202) 326-6400

### **National Council of Teachers of Mathematics**

An Agenda for Action: Recommendations for School Mathematics of the 1980's. 1980. Curriculum and Evaluation Standards for School Mathematics. 1989. Assessment Standards for School Mathematics. 1995.

For further information contact:

National Council of Teachers of Mathematics 1906 Association Drive Reston, VA 22091-1593 (703) 620–9840

### **National Research Council**

### Everybody Counts: A Report to the Nation on the Future of Mathematics Education. 1989.

### National Science Education Standards. 1996.

For further information contact: National Research Council 2101 Constitution Avenue, NW Washington, DC 20418 (202) 334–1399

### **National Science Teachers Association**

Essential Changes in Secondary School Science: Scope, Sequence and Coordination of Secondary School Science. 1989.

For further information contact: National Science Teachers Association 1840 Wilson Boulevard Arlington, VA 22201 (703) 243–7100

Minnesotans have helped create the new vision of mathematics and science education since the first national standards document was released in draft form (NCTM's *Curriculum and Evaluation Standards* in 1987). The early efforts were

largely sponsored by Minnesota Mathematics Mobilization (M<sup>3</sup>), a state coalition

of K–16 teachers, mathematicians, scientists, business, and government; and by the Minnesota Council of Teachers of Mathematics (MCTM). At several state conferences in the late eighties, Minnesotans met with representatives of national organizations to convey their critiques of emerging

## APPENDIX C: THE HIST ORY OF TRANSFORMING TEACHER EDUCATION AND THE ROLE OF SCIMA TH<sup>MN</sup>

documents. A number of Minnesotans were

also involved in creating and critiquing the early science documents, and a statewide conference critiquing the NRC Science Standards was sponsored by SciMath<sup>MN</sup> in January of 1994.

As the vision for student learning of mathematics and science became clear, the attention in Minnesota turned to the vision of teaching and teacher education that would enable the new ways of learning. M<sup>3</sup> sponsored another series of meetings:

a critique of the NCTM *Professional Standards for Teaching* draft in 1990; and two Call For Change in Minnesota conferences in 1992 and 1994 that explored the mathematical preparation needed for elementary and secondary teachers.

In the meantime, a group of Minnesota leaders of science and mathematics were meeting to create SciMath<sup>MN</sup>, a statewide initiative to accelerate the pace, broaden the impact, and increase the effectiveness of K–12 mathematics and science education in Minnesota. The vision of SciMath<sup>MN</sup>, which became a reality in the fall of 1993, is to increase the participation and achievement of all Minnesota students in mathematics and science, and the target of its initiatives are those who shape the opportunities children have for learning and applying knowledge in their adult lives: policymakers, teachers, parents, higher education, and business. A central thrust of the SciMath<sup>MN</sup> Transforming Teacher Education effort is the education and continuing professional development of teachers of mathematics and science.

The SciMath<sup>MN</sup> Teacher Preparation Committee, composed of representatives from K–12, higher education, and business began meeting as part of the process of creating SciMath<sup>MN</sup>. The committee evolved into the Transforming Teacher Education initiative with the following mission:

To begin the process of transforming teacher preparation in mathematics and science so that Minnesota's K–12 teachers are prepared to teach according to the vision of the present and future national mathematics and science education standards, and are prepared to continue learning new content

### and ways of teaching throughout their professional lives.

The first Transforming Teacher Education meeting was held in January 1994 at St. Cloud State University. Participants included representatives from government, business, and the K–12 community, along with scientists, mathematicians, and educators from Minnesota's public and private colleges and universities. The purpose of the conference was "to identify essential elements of knowledge that beginning teachers need to know in order to teach according to current and emerging recommendations for science and mathematics curriculum and teaching." After extensive discussion, those present were able to reach an amazing level of consensus on four dimensions of teacher's knowledge: knowledge of children, schools, content, and pedagogy.

Subsequent Transforming Teacher Education meetings were held in 1994, and 1995. The July 1994 meeting resulted in plans for the next two conferences and for this document. In October 1994 and April 1994, two conferences were held which involved the teacher education community of Minnesota in the process of Transforming Teacher Education. Teams of faculty from each of Minnesota's teacher preparation institutions attended the conferences together. Between the two meetings, teams worked at their institutions to begin the process of examination and revitalization of their programs. In keeping with the basic premise of TTE that the whole university educates the future teacher, institutional teams included representatives from mathematics, science, and teacher educations departments, university administrators and partner K–12 practitioners. For the same reason, teams of community college faculty also participated in the TTE initiative. In

July 1995, small teams of Transforming Teacher Education participants came together to complete a final review of *Transforming Teacher Education: A Minnesota Framework for Mathematics and Science.* 

During the 1995–96 school year, SciMath<sup>MN</sup>'s efforts continue with the publication of this document, two statewide conferences, and with work at 20 Minnesota teacher preparation institutions where dialogue continues regarding the renovation of teacher education in mathematics and science.

### APPENDIX D: MA THEMATICS AND SCIENCE BIBLIOGRAPHIES

Compiled by Dale Pearson

- Abell, Sandra. "Helping Science Methods Students Construct Meaning from Text." *Journal of Science Teacher Education*. 1992, 3(1), pp. 11–15.
- Adams, Verna M. *Teacher Talk: Cognitive Goals Inferred from Instruction.* Apr. 1993. ERIC Microfiche ED364432.
- Alvarez, Lolina, et. al. "Calculus Instruction at New Mexico State University through Weekly Themes and Cooperative Learning." *Primus.* 1993, 3(1), pp. 83–98.
- Ambrosio, Anthony. "The Effects of Cooperative Learning in a Physical Science Course for Elementary/Middle Level Preservice Teachers." *Journal of Research in Science Teaching.* 1993, 30(7), pp. 697–707.
- Anderson, C. "The Role of Education in the Academic Disciplines in Teacher Preparation" in A. E. Woolfolk, ed. *Research Perspectives on the Graduate Preparation of Teachers.* 1989. pp. 88–107.
- Apple, Michael W. "Do the Standards Go Far Enough? Power, Policy and Practice in Mathematics Education." *Journal for Research in Mathematics Education*. 1993. pp. 413–431.
- Arcavi, Abraham, and Alan Schoenfeld. "Mathematics Tutoring Through a Constructivist Lens: The Challenges of Sense–Making." *Journal of Mathematical Behavior.* 1992, 11(4), pp. 321–335.
- Armstrong, James O. Learning To Make Idea Maps with Elementary Science Text. Technical Report No. 572. Apr. 1993. ERIC Microfiche ED355491.
- Austin, Rick. "Mathematics Teaching and Teachers in the Year 2000. 88 Special Theme Issues: Education in the Year 2000." *Clearing House.* 1988, 62(1), pp. 23–25.
- Bagheri, Hooshi, et. al. Restructuring Teacher Education: Integrating Science/Mathematics Methods Course and Student Teaching – the Northridge Experience. Feb. 1991. ERIC Microfiche ED338604.
- Bagley, Theresa. "Assessing Students' Disposition: Using Journal to Improve Students' Performance." *Mathematics Teacher*. 1993, 85(8), pp. 660–663.
- Baird, John R., et. al. "The Importance of Reflection in Improving Science Teaching and Learning." *Journal of Research in Science Teaching.* 1991, 28(2), pp. 163–182.

Baker, Dale R. A Summary of Research in Science Education - 1989. Dec. 1990.

ERIC Microfiche ED335237.

- Ball, Deborah L. Breaking with Experience in Learning To Teach Mathematics: The Role of a Preservice Methods Course: Issue Paper 89–10. Nov. 1989. ERIC Microfiche ED318696.
- Ball, Deborah L. "Halves, Pieces, and Twoths: Constructing Representational Contexts in Teaching Fractions." *Craft Paper.* 1990, 90(2), ERIC Microfiche ED324226.
- Ball, Deborah L. *Unlearning to Teach Mathematics (Issue Paper 88–1).* Michigan State University, National Center for Research of Teacher Education.
- Ball, Deborah L. "Research on Teacher Learning: Studying How Teachers' Knowledge Changes." *Action in Teacher Education.* 1988, 10(2), pp. 17–24.
- Ball, Deborah L., and G. Williamson McDiarmid. "The Subject–Matter Preparation of Teachers" in *Handbook of Research on Mathematics Teaching and Learning*. 1992. pp. 437–449.
- Ball, Deborah L. "The Mathematical Understanding that Prospective Teachers Bring to Teacher Education." *Elementary School Journal.* 1990, 90(4), pp. 449–466.
- Ball, Deborah L., and S. Feiman–Nemster. "Using Textbooks and Teachers' Guides: A Dilemma for Beginning Teachers and Teacher Educators." *Curriculum Inquiry.* 1988, (18), pp. 401–423.
- Ball, Deborah Loewenberg. "What's All This Talk About 'Discourse'?" Arithmetic Teacher. 1991, (11), pp. 44–47.
- Ball, Deborah Loewenberg. "With an Eye on the Mathematical Horizon: Dilemmas of Teaching Elementary School Mathematics." *Craft Paper.* 1990, 90(3), ERIC Microfiche ED324227.
- Barba, Robertta. "A Comparison of Preservice and In–Service Earth and Space Science Teachers' General Mental Abilities, Content Knowledge, and Problem–Solving Skills." *Journal* of Research in Science Teaching. 1992, 29(10), pp. 1021–1035.

Barnes, Sue Jackson. "Involve the Community." *Mathematics Teacher.* 1993, 86(6), pp. 442–448.

Baron, Eleanor. Math and Science Professionals Are Becoming Teachers: The Duquesne Model. Aug. 1990. ERIC Microfiche ED341644.
- Beisenherz, Paul C. "Preparing Secondary Teachers to Study Science Teaching." *Journal of Science Teacher Education.* 1991, 2(2), pp. 40–44.
- Bereiter, C., and M. Scardamalia. "Intentional Learning as a Goal of Instruction" in
  L. B. Resnick, ed. *Knowing, Learning and Instruction: Essays in Honor of Robert Glaser.* 1989. pp. 361–392.
- Berg, Craig A. "Interviewing: Towards Developing the Tools of a Constructivist Teacher." *Journal of Science Teacher Education.* 1993, 4(1), pp. 24–29.
- Beyerbach, Barbara A. "Using a Computerized Concept Mapping Program to Assess Preservice Teachers' Thinking about Effective Teaching." *Journal of Research in Science Teaching.* 1990, 27(10), pp. 961–971.
- Bishop, A. "The Social Construction of Meaning A Significant Development for Mathematics Education?" *For the Learning of Mathematics.* 1985, 5(1), pp. 24–28.
- Bishop, Alan J. *Mathematical Enculturation: A Cultural Perspective on Mathematics Education.* 1991. ERIC Microfiche ED341565.
- Blank, Rolf K. Has Science and Mathematics Education Improved Since "A Nation At Risk?" Trends in Course Enrollments, Qualified Teachers, and Student Achievement. Jan. 1992. ERIC Microfiche ED342679.
- Blosser, Patricia E. National Association for Research in Science Teaching: Annual Conference Proceedings (63rd, Atlanta, Georgia, April 8–11, 1990). April 1990. ERIC Microfiche ED335235.
- Blosser, Patricia E. *Using Cooperative Learning in Science Education.* Jan. 1993. ERIC Microfiche ED351207.
- Blum–Anderson, Judy. "Increasing Enrollment in Higher–Level Mathematics Classes through the Affective Domain." *School Science and Mathematics.* 1992, 92(8), pp. 433–436.
- Bodenhausen, Judith. Using Cognitive Research to Turn a High School 'Remedial' Mathematics Program Inside–Out: A Teacher's Perspective. Apr. 1992. ERIC Microfiche ED355096.
- Borko, H., and C. Livingstone. "Cognition and Improvisation: Differences in Mathematics Instruction by Expert and Novice Teachers." *American Education Research Journal.* 1987, 26(4), pp. 473–498.



- Borko, Hilda, M. Eisenhart, C. A. Brown, R. G. Underhill, D. Jones, and P. C. Agard. "Learning to Teach Hard Mathematics: Do Novice Teachers and Their Instructors Give Up Too Easily?" *Journal for Research in Mathematics Education.* 1992, 23(3), pp. 194–222.
- Brensky, Judith, et. al. "Reasoning Fallacies in Preservice Elementary School Teachers." *Research in Science and Technological Education.* 1992, 10(1), pp. 83–92.
- Broekman, Harrie G. B. "The Prehistory of Teacher Trainees and the Consequences for Teacher Education." Journal of Mathematical Behavior. 1987, 6(2), pp. 201–216.
- Brophy, J., ed. Advances on Research in Teaching. JAI Press. 1989.
- Brown, Catherine A., and Hilda Borko. "Becoming a Mathematics Teacher" in Douglas A. Grouws, ed. *Handbook of Research on Mathematics Teaching and Learning.* 1992. pp. 209–239.
- Brown, Laurinda. "The Influence of Teachers on Children's Image of Mathematics." For the Learning of Mathematics. 1992, 12(2), pp. 29–33.
- Brown, Mary Jo McGee. *Teaching as an Interpretive Inquiry Process.* 1992, 92(10), ERIC Microfiche ED365525.
- Brown, Stephen I., Thomas J. Cooney and Doug Jones. "Mathematics Teacher Education" in M. C. Wittrock, ed. *Handbook of Research on Teaching*, 3rd edition. 1992. pp. 639–657.
- Brutlag, Dan. "Making Connections: Beyond the Surface." *Mathematics Teacher.* 1992, 85(3), pp. 230–235.
- Campbell, Patricia B. Working Together, Making Changes: Working In and Out of School To Encourage Girls in Math and Science. Encouraging Girls in Math and Science Series. 1992. ERIC Microfiche ED350170.
- Cappo, Marge. "Teach Students to Communicate Mathematically." *Computing Teacher.* 1991, 18(5), pp. 34–39.
- Carey, Deborah A. "The Patchwork Quilt: A Context for Problem Solving." *Arithmetic Teacher*. 1992, 40(4), pp. 199–203.
- Carnine, D. "The Missing Link in Improving Schools Reforming Educational Leaders." Direct Instruction News. 1992, 11(3), pp. 25–35.

- Cavallo, Ann Liberatore. *The Retention of Meaningful Understanding of Meiosis and Genetics.* 1992. ERIC Microfiche ED356141.
- Cavallo, Ann Liberatore. *Students Meaningful Learning Orientation and Their Meaningful Understandings of Meiosis and Genetics.* Mar. 1992. ERIC Microfiche ED356140.
- Champagne, A. B. "The Psychological Basis for a Science Teaching Model." Washington, DC. *American Association for the Advancement of Science*. 1987.
- Chaste, Audrey M. "Field Experiences in Secondary Teacher Education: Qualitative Differences and Curriculum Change." *Teaching and Teacher Education.* 1993, 9(2), pp. 7–15.
- Cheng, Yeong–Jing. "Biology Cognitive Preferences of Preservice Biology Teachers." *Proceedings of the National Science Council*, Republic of China. 1991, 1(1), pp. 32–40.
- Clement, John. "Constructivism in the Classroom." *Journal for Research in Mathematics Education.* 1986, 86(7), pp. 422–428.
- Cobb, Paul, et. al. "A Constructivist Alternative to the Representational View of Mind in Mathematics Education." *Journal for Research in Mathematics Education.* 1992, 17(1), pp. 2–33.
- Cobb, Paul, and Leslie Steffe. "The Constructivist Researcher as Teacher and Model Builder." *Journal for Research in Mathematics Education.* 1989, 14(2), pp. 83–94.

Cobb, Paul, E. Yackel, and T. Wood. "Curriculum and Teacher Development: Psychological and Anthropological Perspectives" in Fennema, Elizabeth,
T. Carpenter, and S. J. Lamon. eds. *Integrating Research on Teaching and Learning Mathematics.* 1988. pp. 92–131.

- Cobb, Paul. "A Constructivist Perspective on Information Processing Theories of Mathematical Activity." *International Journal of Educational Research.* 1990, 14(1), pp. 67–92.
- Cobb, Paul. "The Tension Between Theories of Learning and Instruction in Mathematics Education." *Educational Psychologist.* 1991, 23(2), pp. 87–103.
- Cobb, Paul. "Information–Processing Psychology and Mathematics Education A Constructivist Perspective." *Journal of Mathematical Behavior.* 1987, 6(1), pp. 3–40.

Cobb, Paul, T. Wood, and E. Yackel. "Discourse, Mathematical Thinking, and Classroom

Practice" in N. Minick, E. Forman, and A. Stone. eds. *Education and Mind: Institutional, Social, and Developmental Processes*. Oxford University Press, Oxford. 1993.

- Cobern, William W. Contextual Constructivism: The Impact of Culture on the Learning and Teaching of Science. Apr. 1991. ERIC Microfiche ED338488.
- Comfrey, J. "What Constructivism Implies for Teaching." *Journal for Research in Mathematics.* Monograph No. 4. 1990. pp. 107–122.
- Cooney, Thomas J. "The Issue of Reform: What Have We Learned from Yesteryear?" *Mathematics Teacher.* 1988, 81(5), pp. 352–363.
- Copeland, W. "Student Teachers and Cooperating Teachers: An Ecological Relationship." *Theory into Practice.* 1989, 80(18), pp. 194–199.
- Crowley, Mary L. "Student Mathematics Portfolio: More Than a Display Case." *Mathematics Teacher.* 1993, 86(7), pp. 544–547.
- Dai, Meme F. Misconceptions About the Moon Held by Preservice Teachers in Taiwan. 1990. ERIC Microfiche ED325327.
- Davis, Nancy T. "Using Concept Mapping to Assist Prospective Elementary Teachers in Making Meaning." *Journal of Science Teacher Education.* 1990,1(4), pp. 66–69.
- Davis, Robert B., C. A. Maher, and N. Noddings. eds. Constructivist Views on the Teaching and Learning of Mathematics. Monograph No. 4. 1990. Reston, VA. NCTM.
- Davis, Robert B. "Understanding 'Understanding'." *Journal of Mathematical Behavior.* 1992, 11(3), pp. 225–242.
- Davis, Robert B. "The Convergence of Cognitive Science and Mathematics Education." *Journal of Mathematical Behavior.* 1986, 5, pp. 321–333.
- Davis, Robert B. "The Culture of Mathematics and the Culture of Schools." *Journal of Mathematical Behavior.* 1989, 8, pp. 143–160.
- Davis, Robert B. "A Study of The Process of Making Proofs." *Journal of Mathematical Behavior*. 1985, 4(1), pp. 37–41.
- Davis, Robert B., and C. A. Maher. "The Nature of Mathematics: What do we do when we 'Do Mathematics'?" in Robert B. Davis, C. A. Maher, and N. Noddings. eds. *Constructivist Views on the Teaching and Learning of Mathematics.* 1990. pp. 65–78.

- Davis, Robert B. *Learning Mathematics: The Cognitive Science Approach to Mathematics Education.* Ablex Publishing, Reno, Nevada. 1984.
- Davidson, David. "Perspectives on Writing Activities in the Mathematics Classroom." *Mathematics Education Research Journal.* 1990, 2(1), pp. 15–22.
- Dessart, Donald J. A Review and Synthesis of Research in Mathematics Education Reported During 1987. Jan. 1989. ERIC Microfiche ED306143.
- Disinger, John F. *Trends and Issues Related to the Preparation of Teachers for Environmental Education. Environmental Education Information Report.* Dec. 1990. ERIC Microfiche ED335233.
- Doherty, Paul. Hands–On Science: A Teacher's Guide to Student–Built Experiments and The Exploratorium Science Snackbook. 1992. ERIC Microfiche ED362410.
- Donmoyer, Robert, et. al. "The Knowledge and Pedagogical Base of Science Education: An Overview." *Teaching Education.* 1991, 3(2), pp. 11–16.
- Doyle, W. "Work in Mathematics Classes: The Context of Students' Thinking During Instruction." *Educational Psychologist.* 1988, 23, pp. 167–180.
- *Earth Science Education for the 21st Century: A Planning Guide.* Feb. 1991. ERIC Microfiche ED340596.
- Edgerton, Richard T. A Description of the Assessment Practices of Teachers Who Have Begun to Implement the Instructional Practices Suggested in the NCTM Standards Documents. 23 Apr. 1992. ERIC Microfiche ED350162.
- Edgerton, Richard T. The Emerging Implementer of the NCTM "Standards:" Concepts Searching for Expression. Apr. 1993. ERIC Microfiche ED358140.
- Eisenhart, Margaret, et. al. "Conceptual Knowledge Falls through the Cracks: Complexities of Learning to Teach Mathematics for Understanding." *Journal for Research in Mathematics Education.* 1993, 24(1), pp. 8–40.
- Eisenhart, Margaret, L. Behm, and L. Romagnano. "Learning to Teach: Developing Expertise or Rite of Passage?" *Journal for Education for Teaching.* 1991, 17, pp. 51–71.
- Ellerton, Nerida. "Some Pluses and Minuses of Radical Constructivism in Mathematics Education." *Mathematics Education Research Journal.* 1992,

4(2), pp. 1-22.

- Emidio–Caston, Marianne. *Simulation and Meta Processing: Affective Component of Math Procedures.* 27 Feb 1993. ERIC Microfiche ED360258.
- Engelhardt, Jon M. "Focus on Teacher Education in Diagnostic and Prescriptive Mathematics: A Developmental Constructivist Model of Teacher Education and Diagnostic/Prescriptive Mathematics." *Focus on Learning Problems in Mathematics.* 1988, 10(3), pp. 59–67.
- Even, Ruhama. "Subject Matter Knowledge for Teaching and the Case of Functions." *Educational Studies in Mathematics.* 1990, 21(6), pp. 521–544.
- Farrell, Margaret A. "Implementing the Professional Standards for Teaching Mathematics: Learning from Your Students." *Mathematics Teacher.* 1992, 85(8), pp. 656–659.
- Feiman–Nemser, S., and M. Buchmann. "When is Student Teaching Teacher Education?" *Teaching and Teacher Education.* 1987, 87(3), pp. 255–273.
- Fennema, Elizabeth, and Megan Loef Franke. "Teachers' Knowledge and Its Impact" in Douglas A. Grouws, ed. *Handbook of Research on Mathematics Teaching* and Learning. 1992. pp. 147–164.
- Fennema, Elizabeth, et. al. "Using Children's Mathematical Knowledge in Instruction." *American Educational Research Journal.* Fall 1993, 30(3), pp. 555–583.
- Fennema, Elizabeth, T. P. Carpenter, and P. L. Peterson. "Learning Mathematics with Understanding: Cognitively Guided Instruction" in J. Brophy, ed. *Advances in Research on Teaching.* 1989. pp. 195–212.
- Fennema, Elizabeth, T. Carpenter, and S. J. Lamon. eds. *Integrating Research on Teaching and Learning Mathematics.* Wisconsin Center for Educational Research. Eau Claire, WI, 1988.
- Fey, J. "Mathematics Teaching Today: Perspectives from Three National Surveys." The Mathematics Teacher. 1992, 72, pp. 490–504.
- Finson, Kevin D. "Rural Science Teacher Preparation: Component of the Educational System." Journal of Science Teacher Education. 1990, 1(3), pp. 46–48.
- Fleury, Stephen C. "Educating Elementary Science Teachers: Alternative Conceptions of the Nature of Science." *Teaching Education.* 1991, 3(2), pp. 57–67.

- Frank, Martha L. "What Myths about Mathematics Are Held and Conveyed by Teachers?" *Arithmetic Teacher.* 1990, 37(5), pp. 10–12.
- Friel, Susan N. "Implementing the Professional Standards for Teaching Mathematics: Teachers Building on Students' Thinking." *Arithmetic Teacher.* 1992, 39(7), pp. 32–37.
- Gardner, April L. Critical Issues in Reforming Elementary Teacher Preparation in Mathematics and Science. Conference Proceedings (Greeley, Colorado, October 10–13, 1991). Oct. 1991. ERIC Microfiche ED359176.
- Garofalo, Joe, and David Mtetwa. "Mathematics as Reasoning." *Arithmetic Teacher.* 1990, 37(3), pp. 16–18.
- Garofalo, Joe. "Number–Consideration Strategies Students Use to Solve Word Problems. Focus on Learning Problems in Mathematics." *Arithmetic Teacher.* 1992, 14(2), pp. 37–50.
- Geddert, Phyllis. "Student Success through Outcome–Based Education." *Alberta Journal of Educational Research.* 1993, 39(2), pp. 205–215.
- Gess-Newsome, Julie. "Preservice Biology Teachers' Knowledge Structures as a Function of Professional Teacher Education: A Year–Long Assessment." *Science Education.* 1993, 77(1), pp. 25–45.
- Glaser, R. "Education and Thinking: The Role of Knowledge." *American Psychologist.* 1984, 39, pp. 93–104.
- Glick, Judith Gail, et. al. Sources Used by Student Teachers in Lesson Planning. 1992. ERIC Microfiche ED350289.
- Gold, Eric R. "Math: A Journey to Find Sense and Meaning for Students." *Hands On.* 1992, 43–44, pp. 84–86.
- Goldenberg, E. P. "Seeing Beauty in Mathematics: Using Fractal Geometry to Build a Spirit of Mathematical Inquiry." *Journal of Mathematical Behavior.* 1989, 8(3), pp. 169–204.
- Gordon, Myles. *What Is the 'Geometric Supposer' a Case of? Reports and Papers in Progress.* Report No. 90–95. Dec. 1990. ERIC Microfiche ED 344750.
- Gorman, Jacqueline, et. al. *Preparing Elementary Teachers to Teach Mathematics: A Problem–Solving Approach. Final Report. Volume IV: School Components.* Jun. 1992. ERIC Microfiche ED349183.

- Gray, Susan S. "Ideas in Practice: Metacognition and Mathematical Problem Solving." *Journal* of *Developmental Education.* 1991, 14(3), pp. 24–26.
- Greeno, J. G. "A Perspective on Thinking." *American Psychologist.* 1989, 44, pp. 134–141.
- Grossman, P. L., S. M. Wilson, and L. S. Shulman. "Teachers of Substance: Subject Matter Knowledge for Teaching" in M. C. Reynolds. ed. *Knowledge Base for the Beginning Teacher*. 1989. pp. 23–36.
- Grouws, Douglas A., Thomas J. Cooney, and Douglas Jones. eds. *Perspectives on Research on Effective Mathematics Teaching.* NCTM, Washington, DC. 1988.
- Hall, Donald A. "The Influence of an Innovative Activity–Centered Biology Program on Attitudes toward Science Teaching among Preservice Elementary Teachers." *School Science* and Mathematics. 1992, 92(5), pp. 239–242.
- Hart, Lynn C., and Mareene Estes. "Mathematics Staff Development from a Constructivist Perspective." *Journal of Staff Development.* 1990, 11(3), pp. 8–10.
- Harty, Harold, et. al. "Understanding the Nature of Science and Attitudes toward Science and Science Teaching of Preservice Elementary Teachers in Three Preparation Sequences." *Journal of Elementary Science Education.* 1991, 3(1), pp. 13–22.
- Heikkinen, Henry W., et. al. "Classroom Teachers as Agents of Reform in University Teacher Preparation Programs." *Journal of Teacher Education.* 1992, 43(4), pp. 21–36.
- Helgeson, Stanley L. *Trends and Issues in Science Education.* Jun. 1992. ERIC Microfiche ED357954.
- Hembree, Ray. "Experiments and Relational Studies in Problem Solving: A Meta–Analysis." Journal for Research in Mathematics Education. 1992, 23(3), pp. 242–273.
- Hennessey, M. Gertrude. "Students' Ideas about Their Conceptualization: Their Elicitation through Instruction." Apr. 1993. ERIC Microfiche ED361209.
- Herrington, Tony, et. al. "The Ideal Teacher." *Australian Mathematics Teacher.* 1992, 48(1), pp. 4–7.

Hiebert, James, and Thomas P. Carpenter. "Learning and Teaching with Understanding" in

Douglas A. Grouws, ed. *Handbook of Research on Mathematics Teaching and Learning.* 1992. pp. 65–97.

- Hiebert, James. "A Theory of Developing Competence with Written Mathematical Symbols." *Educational Studies in Mathematics.* 1988, 19, pp. 333–355.
- Hiebert, James, and D. Wearne. "Instruction and Cognitive Change in Mathematics." *Educational Psychologist.* 1988, 23, pp. 105–117.

Hiebert, James. ed. *Conceptual and Procedural Knowledge: The Case of Mathematics.* Lawrence Earlbaum Associates, New York, New York, 1986.

- Hodge, Evelyn A. I *ntervention for At–Risk Student at the Secondary Level.* 1991. ERIC Microfiche ED339764.
- Hofmeister, Alan. "Elitism and Reform in School Mathematics." *Remedial and Special Education.* 1993, 14(6), pp. 1–13.
- Hojnacki, Susan K. Thinking Mathematics: What's in It for the Students? Apr. 1992. ERIC Microfiche ED355095.
- Howe, Robert W., et. al. *Trends and Issues in Mathematics Education: Curriculum and Instruction.* Dec. 1990. ERIC Microfiche ED335231.
- Hynd, Cynthia R., et. al. Prospective Teachers' Comprehension and Teaching of a Complex Science Concept. Reading Research Report No. 4. 1993. ERIC Microfiche ED361685.

Jungck, John R. "Constructivism, Computer Exploratoriums, and Collaborative Learning: Constructing Scientific Knowledge." *Teaching Education.* 1991, 3(2), pp. 151–170.

Karp, Karen Silliman. Elementary School Teachers' Attitudes Toward Mathematics: Impact on Students' Autonomous Learning Skills. Mar. 1989. ERIC Microfiche ED307156.

Kieren, Thomas E. "Understanding for Teaching for Understanding." The Alberta Journal of Educational Research. 1990, 36(3), pp. 191–201.

King, Bruce B. "Beginning Teachers' Knowledge of and Attitudes Toward History and Philosophy of Science." *Science Education.* 1991, 75(1), pp. 135–141.

Kirchhoff, Susan. Collaborative University/School District Approaches for Student Teaching Supervision. Nov. 1989. ERIC Microfiche ED314389.

- Kleinfeld, Judith. Preparing Prospective Teachers To Develop the Mathematical and Scientific Abilities of Young Women: The Development of Teaching Cases. Final Report. Jul. 1991. ERIC Microfiche ED346025.
- Kloosterman, Peter, et. al. *Preparing Elementary Teachers to Teach Mathematics: A Problem–Solving Approach.* Jun. 1992. ERIC Microfiche ED349184.
- Kober, Nancy. What We Know about Mathematics Teaching and Learning. Jun. 1991. ERIC Microfiche ED343793.
- Koch, Janice. "Face to Face with Science Misconceptions." *Science and Children.* 1993, 30(6), pp. 39–40.
- Koch, Laura Coffin. "Revisiting Mathematics." *Journal of Developmental Education.* 1992, 16(1), pp. 12–14, 16, 18.
- Koretz, Daniel. "Arriving in Lake Wobegon: Are Standardized Tests Exaggerating Achievement and Distorting Instruction?" *American Educator.* 1988, 12(2), pp. 8–15, 46–52.
- Korthagen, Fred A. J. Characteristics of Reflective Practitioners: Towards an Operationalization of the Concept of Reflection. Apr. 1991. ERIC Microfiche ED334183.
- Kyriacou, Chris. "Small Group Work in Secondary School Mathematics." *Mathematics in School.* 1991, 20(3), pp. 44–46.
- Lacampagne, Carole B. *State of the Art: Transforming Ideas for Teaching and Learning Mathematics.* Jul. 1993. ERIC Microfiche ED360188.
- Lampert, Mandalene. "When the Problem Is Not the Question and the Solution Is Not the Answer: Mathematical Knowing and Teaching." *American Educational Research Journal*. 1990, 27(1), pp. 29–63.
- Lampert, Mandalene. "Connecting Mathematical Teaching and Learning" in E. Fennema, T. P. Carpenter, and S. J. Lamon. eds. *Integrating Research on Teaching and Learning Mathematics.* 1988. pp. 132–167.

Lapointe, Archie F., et. al. Learning Mathematics. Feb. 1992. ERIC Microfiche ED347081.

Lappan, Glenda and Joan Ferrini–Mundy. "Knowing and Doing Mathematics: A New Vision for Middle Grades Students." *The Elementary School Journal.* 1993, 93(5), pp. 625–641.

- Lappan, Glenda. "What Do We Have and Where Do We Go from Here?" *Arithmetic Teacher.* May 1993, 40(3), pp. 524–526.
- Latz, Mark. "Preservice Teachers' Perceptions and Concerns about Classroom Management and Discipline: A Qualitative Investigation." *Journal of Science Teacher Education.* 1992, 3(1), pp. 1–4.
- Lavoie, Derreck R. The Construction and Application of a Cognitive–Network Model of Prediction Problem Solving. Apr. 1991. ERIC Microfiche ED332864.
- Leder, Gilah. Assessment and Learning Mathematics. Sep. 1992. ERIC Microfiche ED362413.
- Leder, Gilah. "Is Teaching Learning?" Australian Mathematics Teacher. 1991, 1, pp. 4–7.
- Leder, Gilah C. "Constructivism: Theory for Practice?" *The Case of Mathematics. Higher Education Research and Development.* 1993, 12(1), pp. 5–20.
- Lederman, Norman G., et. al. Becoming a Teacher: Balancing Conceptions of Subject Matter and Pedagogy. Apr. 1993. ERIC Microfiche ED361294.
- Leinhart, G., R. T. Putnam, M. K. Stein, and J. Baxter. "Where Subject Knowledge Matters" in J. E. Brophy, ed. Advances in Research on Teaching: Teachers' Subject Matter Knowledge and Classroom Instruction. 1991. Vol 2. pp. 87–113.
- Leinhart, G., and J. C. Greeno. "The Cognitive Skill of Teaching." *Journal of Educational Psychology.* 1986, 12(2), pp. 75–95.
- Lester, Frank K., Jr., et. al. Preparing Elementary Teachers to Teach Mathematics: A Problem-solving Approach. Final Report. Volume II: Content Component. Jun. 1992. ERIC Microfiche ED349181.
- Lester, Frank K., Jr. "Teaching Mathematics via Problem Solving: A Course for Prospective Elementary Teachers." *For the Learning of Mathematics.* 1993, 13(2), pp. 8–11.
- Levine, Deborah R. "Tips for Beginners." Mathematics Teacher. 1991, 84(5), pp. 454–456.
- Lewellen, Hester. "Transforming Yourself as a Teacher: Is It Possible?" *Ohio Journal* of *School Mathematics.* 1992, 24, pp. 1–10.
- Lindquist, Mary M. "A Bold New Vision in Mathematics Education." *Momentum.* 1993, 4(3), pp. 7–9.
- Livingstone, C., and H. Borko. "High School Mathematics Review Lessons: Expert–novice Distinctions." *Journal for Research in Mathematics Education.* 1990, 21(5), pp. 372–387.

- Lochhead, Jack. "Knocking Down the Building Blocks of Learning: Constructivism and the Vantures Program." *Educational Studies in Mathematics.* 1992, 23(5), pp. 543–552.
- Madsen, Anne I. The Effects of Conceptually Oriented Instruction on Students' Computational Competencies. Aug. 1992. ERIC Microfiche ED351194.
- Maor, Dorit. *Development of Student Inquiry Skills: A Constructivist Approach in a Computerized Classroom Environment.* Apr. 1991. ERIC Microfiche ED336261.
- Marks, Rick. "Pedagogical Content Knowledge: From a Mathematical Case to a Modified Conception." *Journal of Teacher Education.* 1990, 41(3), pp. 3–11.
- Mathematical Sciences Education Board. *Reshaping School Mathematics.* National Academy Press, Washington, DC., 1990.
- Mathematical Sciences Education Board. *On the Shoulders of Giants*. National Academy Press, Washington, DC., 1990.
- Mayer, Richard E. "Special Section: Cognition and Instruction in Mathematics." *Journal of Educational Psychology.* 89, 81(4), pp. 452–593.
- McCoy, Leah P. "Correlates of Mathematics Anxiety. Focus on Learning Problems in Mathematics." 1992, 14(2), pp. 51–57.
- McDiarmid, and G. Williamson. "An Exploration of the Subject Matter Knowledge of Alternative Route Teachers: Can We Assume They Know Their Subject?" *Journal of Teacher Education.* 1991, 42(2), pp. 93–103.
- McDiarmid, G. W., D. L. Ball, and C. P. Anderson. "Why Staying Ahead One Chapter Just Won't Work: Subject–Specific Pedagogy" in M. Reynolds, ed. *Knowledge Base for the Beginning Teacher.* New York. Pergamon Press. 1989.
- McMillan, Thomas C. "Hypocycloids: A Student Exercise." *Mathematics and Computer Education.* 1992, 26(2), pp. 184–194.
- McNamara, David. "Can Research Inform Classroom Practice? The Particular Case of Buggy Algorithms and Subtraction Errors." *Teaching and Teacher Education.* 1991, 7(4), pp. 395–403.

McNeill, Ruth. "A Reflection on When I Loved Math and How I Stopped." Journal of

Mathematical Behavior. 1988, 7(1), pp. 45-50.

- McQualter, J. W. "Becoming a Mathematics Teacher: The Use of Personal Construct Theory." *Educational Studies in Mathematics.* 1986, 17(1), pp. 1–14.
- Mercer, Cecil D., C. A. Harris, and S. P. Miller. "First Invited Response: Reforming Reforms in Mathematics." *Remedial and Special Education.* 1993, 14(6), pp. 14–19.
- Merseth, Katherine K. "How Old Is the Shepherd? An Essay about Mathematics Education." *Phi Delta Kappan.* 1993, 74(7), pp. 548–554.
- Meserve, Bruce E. "Looking Ahead in Teacher Preparation: A Personal Perspective on NCTM–MAA Cooperation." *Mathematics Teacher.* 1989, 82(7), pp. 564–570.
- Mestre, Jose. "Why Should Mathematics and Science Teachers Be Interested in Cognitive Research Findings?" *Academic Connections.* The College Board. 1987. pp. 3–5, 8–11.
- Middleton, James A. *Teachers' vs. Students' Beliefs Regarding Intrinsic Motivation in the Mathematics Classroom: A Personal Constructs Approach.* Apr. 1992. ERIC Microfiche ED353154.
- Midkiff, Ruby Bostick, et. al. *Learning Style Needs of At–Risk Students: Teaching Math and Social Studies the Way They Learn.* 1991. ERIC Microfiche ED331632.
- Moore, Alan D. *How Do Teachers Feel about Science? Measurement of Attitudes Towards Science.* Oct. 1990. ERIC Microfiche ED325511.
- Mosquera, Julio C. "Advanced Mathematics from an Elementary Standpoint." *Mathematics Educator.* 1992, 3(1), pp. 11–15.
- Munby, Hugh. *The Authority of Experience in Learning To Teach: Messages from a Physics Methods Class.* Apr. 1993. ERIC Microfiche ED362475.
- National Council of Teachers of Mathematics. Commission on Professional Standards for Teaching Mathematics. *Professional Standards for Teaching Mathematics*.
   Reston, VA: The Council. 1991.
- National Council of Teachers of Mathematics. Commission on Standards for School Mathematics. *Curriculum and Evaluation Standards for School Mathematics.* Reston, VA: The Council. 1989.
- National Research Council. Reshaping School Mathematics: A Philosophy and Framework for Curriculum. National Academy Press, Washington, DC., 1990.

National Research Council. *Everybody Counts: A Report to the Nation on the Future* of Mathematics Education. National Academy Press, Washington, DC., 1989.

New Trends in Integrated Science Teaching, Volume VI. 1990. ERIC Microfiche ED347054.

- Niaz, Mansoor. Teaching Algorithmic Problem Solving or Conceptual Understanding: Role of Developmental Level, Mental Capacity, and Cognitive Style. Apr. 91. ERIC Microfiche ED331717.
- Nicholls, John G., et. al. "Assessing Students' Theories of Success in Mathematics: Individual and Classroom Differences." *Journal for Research in Mathematics Education.* 1990, 21(2), pp. 109–122.
- Noddings, N. "Constructivism in Mathematics Education" in R. B. Davis, C. A. Mahar, and N. Noddings. eds. *Constructivist Views on the Teaching and Learning of Mathematics.* 1990. pp. 7–18.
- Oakes, Jeannie, et. al. *Multiplying Inequalities: The Effects of Race, Social Class, and Tracking on Opportunities to Learn Mathematics and Science.* Jul. 1990. ERIC Microfiche ED329615.
- O'Connor, James E. *Evaluating the Effects of Collaborative Efforts To Improve Mathematics and Science Curricula.* Apr. 1993. ERIC Microfiche ED357083.
- Ohlsson, Stellan. "Simulating the Understanding of Arithmetic: A Response to Schoenfeld." *Journal for Research in Mathematics Education.* 1992, 23(5), pp. 474–482.
- O'Neil, John. Raising Our Sights: Improving U.S. Achievement in Mathematics and Science. 1991. ERIC Microfiche ED342651.
- Onslow, Barry, et. al. "Developing a Teaching Style: A Dilemma for Student Teachers." *Alberta Journal of Educational Research.* 1992, 38(4), pp. 301–315.
- Owens, John E. "Access to Tools: Preparing Secondary Mathematics Teachers for Classroom Technology." *Journal of Computers in Mathematics and Science Teaching.* 1989, 9(2), pp. 23–30.
- Owens, John E. "A Bridge Too Far: On the Nature of Teacher Training." *For the Learning of Mathematics.* 1992, 12(1), pp. 48–51.
- Parker, O. Jerry, et. al. "Improving Science Instruction: A Collaborative Curriculum." *Journal of Chemical Education.* 1990, 67(4), pp. 327–329.
- Parker, Ruth E. "Implementing the Curriculum and Evaluation Standards: What Will Implementation Take?" *Mathematics Teacher.* 1991, 84(6), pp. 442–449.

- Parsons–Chatman, Sharon. Making Sense of Constructivism in Preservice: A Case Study. 1990. ERIC Microfiche ED319600.
- Patzelt, Karen E. Increasing Homework Completion through Positive Reinforcement. Apr. 1991. ERIC Microfiche ED343306.
- Peck, Donald M. Developing a Pedagogical Useful Content Knowledge in Elementary Mathematics. April 1991. ERIC Microfiche ED332875.

Pedersen, J. E. "The Effects of Hands–On, Minds–On Teaching Experiences on Attitudes of Preservice Elementary Teachers." *Science Education.* 1992, 76(2), pp. 141–146.

Perkins, D. N., and G. Salomon. "Are Cognitive Skills Context Bound?" *Educational Researcher*. 1989, 18(1), pp. 16–25.

- Piburn, Michael D. "Teaching a Hypothesis Testing Strategy to Prospective Teachers." *Journal of Science Teacher Education.* 1992, 3(2), pp. 42–46.
- Piel, John A. *Educational Attitudes of Preservice Teachers, or 'Redesigning the Edsel'* of *Teacher Education.* 1992. ERIC Microfiche ED343897.
- Pirie, Susan, and Thomas Kieren. "Watching Sandy's Understanding Grow." Journal of Mathematical Behavior. 1992, 11(3), pp. 243–257.

Pirie, Susan. "Creating Constructivist Environments and Constructing Creative Mathematics." *Educational Studies in Mathematics.* 1992, 23(5), pp. 505–528.

Pirie, Susan. "Understanding: Instrumental, Relational, Intuitive, Constructed, Formalised?" For the Learning of Mathematics. 1988, 8(3), pp. 2–6.

Platt, M. L. "Short Essay Topics for Calculus." Primus. 1993, 3(1), pp. 42-46.

Powell, Arthur B. "Beyond Questions and Answers: Prompting Reflections and Deepening Understanding of Mathematics Using Multiple–Entry Logs." *For the Learning of Mathematics*. 1992, 12(2), pp. 12–18.

Prawat, R. "Teaching for Understanding: Three Key Attributes." *Teacher and Teacher Education.* 1989, 5(4), pp. 315–328.

Putnam, Ralph T. Learning To Attend to Students' Mathematical Thinking: Case Study of a Collaboration. Elementary Subjects Center Series, No. 79. Jan. 1993. ERIC Microfiche ED355119.

- Rasch, Katharine, et. al. *Mathematical Literacy To Empower Teacher Education Students in the 21st Century: How Can This Become Reality?* Jun. 1992. ERIC Microfiche ED351291.
- Ratay, Gabriella M. "Student Performance with Calculus Reform at the United States Merchant Marine Academy." *Primus.* 1993, 3(1), pp. 107–111.
- Rech, Janice, et. al. "Comparisons of Mathematical Competencies and Attitudes of Elementary Education Majors with Established Norms of a General College Population." *School Science and Mathematics.* 1993, 93(3), pp. 141–144.
- Reiter, Paula J. "Elementary Science Teacher Education: Seven Exemplars." Journal of Science Teacher Education. 1991, 2(1), pp. 22–27.
- Remillard, Janine. Abdicating Authority for Knowing: A Teacher's Use of An Innovative Mathematics Curriculum. Elementary Subjects Center Series, No. 42. Jul. 1991. ERIC Microfiche ED341549.
- Research, Issues, and Practices. Annual Curriculum and Instruction Research Symposium Conference Proceedings. Apr. 1993. ERIC Microfiche ED363607.
- Resnick, Lauren B. "Mathematics and Science Learning: A New Conception." *Science.* 1983, 4(29), pp. 477–478.

Resnick, Lauren B. "Learning In and Out of School." Educational Researcher. 1987, 16(9), pp. 13-20.

- Resnick, Lauren B. ed. *Knowing, Learning and Instruction: Essays in Honor of Robert Glaser.* Lawrence Earlbaum Associate, New York, New York, 1989.
- Rivkin, Mary S. "'Feeling Is First' In ECE Science and Mathematics." *Teaching Education.* 1991, 3(2), pp. 171–176.
- Roberts, Douglas A. "Absorption, Refraction, Reflection: An Exploration of Beginning Science Teacher Thinking." *Science Education.* 1990, 74(2), pp. 197–224.
- Roberts, Douglas A. "What Counts as an Explanation for a Science Teaching Event?" *Teaching Education*. 1991, 3(2), pp. 69–87.
- Romberg, Thomas A. "Further Thoughts on the Standards: A Reaction to Apple." *Journal for Research in Mathematics Education.* 1993. pp. 432–437.
- Romberg, Thomas A. "Problematic Features of the School Mathematics Curriculum" in P. W. Jackson. ed. *Handbook of Research on Curriculum*. 1992. pp. 449–788.

Romberg, Thomas A., and Thomas P. Carpenter. "Research on Teaching and Learning Mathematics: Two Disciplines of Scientific Inquiry" in M. C. Wittrock. ed. *Handbook of Research* on *Teaching*, 3rd edition. 1993. pp. 437–449.

Rosenthal, Dorothy B. "A Learning Cycle Approach to Dealing with Pseudoscience Beliefs of Prospective Elementary Teachers." *Journal of Science Teacher Education.* 1993, 4(2), pp. 33–36.

Rosnick, Peter, and John Clement. "Learning Without Understanding: The Effect of Tutoring Strategies on Algebra Misconceptions." *Journal of Mathematical Behavior.* 1980, 3(1), pp. 3–27.

Rubba, Peter A., et. al. *Excellence in Educational Teachers of Science. The 1993 Yearbook of the Association for the Education of Teachers of Science.* Jan. 1993. ERIC Microfiche ED355111.

Saxe, G. B. "Culture and Cognitive Development: Studies in Mathematical Understanding." Lawrence Erlbaum Associates, Hillsdale, NJ, 1991.

Schmalz, Rosemary. "Problem Solving — An Attitude as Well as a Strategy." Mathematics Teacher. 1989, 82(9), pp. 685–687.

Schmidt, William W. Teachers' and Teacher Candidates' Beliefs about Subject Matter and about Teaching Responsibilities. Feb. 1990. ERIC Microfiche ED320902.

Schmittau, Jean. "Mathematics Education in the 1990s: Can It Afford to Ignore Its Historical and Philosophical Foundations?" *Educational Theory.* 1991, 41(2), pp. 121–133.

Schoenfeld, Alan. "When Good Teaching Leads to Bad Resuts: The Disasters of 'Well Taught' Mathematics Classes." *Educational Psychologist.* 1988. pp. 145–166.

Schoenfeld, Alan. "Learning to Think Mathematically: Problem Solving, Metacognition, and Sense Making in Mathematics" in Douglas A. Grouws, ed. *Handbook of Research on Mathematics Teaching and Learning.* 1992. pp. 334–370

Schoenfeld, Alan H. ed. Cognitive Science and Mathematics Educaton. Lawrence Erlbaum Associates, Hillsdale, NJ, 1987.

Schoenfeld, Alan H. "What's in a Model? Issues in the Use of Simulation Models to Analyze Student Understanding: A Reaction to Ohlsson, Ernst, and Rees." *Journal for Research in Mathematics Education.* 1992, 23(5), pp. 468–473

Schoenfeld, Alan H. "On Having and Using Geometric Knowledge" in J. Hiebert. ed. *Conceptual* and Procedural Knowledge: The Case of Mathematics. 1986. pp. 225–264.

82 Appendix D

- Schulman, Lee. "Those Who Understand: Knowledge Growth in Teaching." *Educational Researcher*. 1986, 15(2), pp. 4–14.
- Schulman, Lee. "Conceptual Knowledge Falls through the Cracks: Complexities of Learning to Teach Mathematics for Understanding." *Journal for Research in Mathematics Education.* 1993, 24(1), pp. 8–40.
- Shannon, Kathleen M. "Special Problems: An Alternative to Student Journals in Mathematics Courses." *Primus.* 1992, 2(3), pp. 247–256.
- Shuell, Thomas J. "The Two Cultures of Teaching and Teacher Preparation." *Teaching and Teacher Education.* 1992, 8(1), pp. 83–90.
- Sigurdson, Sol E., and Anton Olson. "Teaching Mathematics with Meaning." *Journal of Mathematical Behavior*. 1992, 11(1), pp. 37–57.

Simmons, Patricia E. "Exploring Jurassic Park." Science Teacher. 1993, 60(8), pp. 50–53.

- Simon, Martin A. *Learning Mathematics and Learning To Teach: Learning Cycles in Mathematics Teacher Education.* Apr. 1992. ERIC Microfiche ED349174.
- Simon, Martin A. Reconstructing Mathematics Pedagogy from a Constructivist Foundation. Apr. 1993. ERIC Microfiche ED364406.
- Sowder, Judith, et. al. Understanding as a Basis for Teaching: Mathematics and Science for Prospective Middle School Teachers. Final Report. May 1991. ERIC Microfiche ED339590.
- Stacy, Kaye. "Mathematical Problem Solving in Groups: Are Two Heads Better than One?" Journal of Mathematical Behavior. 1992, 11(3), pp. 261–275.
- Stahl, Abraham. "The Interference of Traditional Beliefs and Concepts in the Study of Science." *Journal of Science Teacher Education.* 1992, 3(1), pp. 5–10.

Steen, L. "The Science of Patterns." Science. 1988, 240(29), pp. 611–616.

- Steffe, Leslie P., and Heide G. Wiegel. "On Reforming Practice in Mathematics Education." *Educational Studies in Mathematics.* 1992, 23(5), pp. 445–465.
- Steffe, Leslie P. "Inconsistencies and Cognitive Conflict: A Constructivist View." *Focus on Learning Problems in Mathematics.* 1990, 12(3), pp. 99–109.
- Steffe, Leslie P., and Terry Wood. *Transforming Children's Mathematics Education: International Perspectives.* Lawrence Erlbaum Associates, Hillsdale, NJ, 1990.

- Stenmark, Jean Kerr. ed. Mathematics Assessment: Myths, Models, Good Questions, and Practical Suggestions. 1991. ERIC Microfiche ED345943.
- Stewart, Margaret E. Writing To Learn Mathematics: The Writer Audience Relationship. Mar. 1992. ERIC Microfiche ED347549.
- Stiff, Lee V., et. al. "Using Symbolic Calculators in a Constructivist Approach to Teaching Mathematics of Finance." *Journal of Computers in Mathematics and Science Teaching.* 1992, 11(1), pp. 75–84.
- Stoddart, Trish, et. al. "Reconstructing Elementary Teacher Candidates' Understanding of Mathematics and Science Content." *Teaching and Teacher Education.* 1993, 9(3), pp. 223–241.
- Stoddart, Trish. "Who Is Prepared to Teach in Urban Schools?" *Education and Urban Society.* 1993, 26(1), pp. 29–48.
- Stolworthy, Reed L. Methodological Applications of Theory to Practice by Preservice Secondary School Mathematics Teachers. Oct. 1991. ERIC Microfiche ED342721.
- Strawitz, Barbara M. "The Effects of Review on Science Process Skill Acquisition." Journal of Science Teacher Education. 1993, 4(2), pp. 54–57.
- Suter, Larry E. Indicators of Science & Mathematics Education 1992. First Edition. 1992. ERIC Microfiche ED365511.
- Tankersley, Karen. "Teaching Math Their Way." Educational Leadership. 1993, 50(8), pp. 12–13.
- Tharp, Marcia L. A Problem–Solving Inquiry–Oriented Approach to Learning Mathematics: Student/Teacher Interactions of Rule–Based Learners. Apr. 1992. ERIC Microfiche ED355090.
- Thipkong, Siriporn. "Preservice Elementary Teachers' Misconceptions in Interpreting and Applying Decimals." *School Science and Mathematics.* 1991, 91(3), pp. 93–99.
- Thompson, Alba G. "Assessing Students' Learning to Inform Teaching: The Message in NCTM's Evaluation Standards." *Arithmetic Teacher.* 1989, 37(4), pp. 22–26.
- Tirosh, Dina. "The Effect of Problem Type and Common Misconceptions on Preservice Elementary Teachers' Thinking about Division." *School Science and Mathematics.* 1991, 91(4), pp. 157–163.

Tobias, Randolf. Nurturing At-Risk Youth in Math and Sciences: Curriculum and Teaching Considerations.

1992. ERIC Microfiche ED351137.

- Tobias, Sheila. *They're Not Dumb, They're Different Stalking the Second Tier.* 1990. ERIC Microfiche ED331702.
- Tomic, Welko. "Teaching Behavior and Student Learning Outcomes in Dutch Mathematics Classrooms." *Journal of Educational Research.* 1989, 82(6), pp. 339–347.
- Tomkiewicz, Warren C. *Reflective Teaching and Conceptual Change in an Inter-disciplinary Elementary Methods Course.* Apr. 1991. ERIC Microfiche ED339689.
- Tooke, D. James. "Student Teachers' Mathematical Backgrounds and Attainment of Their Secondary Students." *Clearing House.* 1003, 66(5), pp. 273–277.
- Trent, John H. "Needed: More Better Prepared Junior High School Mathematics Teachers." *School Science and Mathematics.* 1987, 87(2), pp. 100–107.
- Trumbull, Deborah J. "Education 301: Knowing and Learning in Science and Mathematics." *Teaching Education.* 1991, 3(2), pp. 145–150.
- Trumbull, Deborah. "Learning to Ask, Listen, and Analyze; Using Structured Interviewing Assignments to Develop Reflection in Preservice Science Teachers." *International Journal of Science Education.* 1991, 13(2), pp. 129–142.
- Tuan, Hsiao–Lin. The Influence of Preservice Secondary Science Teachers' Beliefs about Science and Pedagogy on Their Planning and Teaching. Apr. 1991. ERIC Microfiche ED332871.
- Turner, Tony. "Preparing Teachers for Science in Multicultural Schools: A Survey of Practice in Initial Teacher Education." *Research in Science and Technological Education.* 1993, 11(2), pp. 157–170.
- Underhill, Robert G. North American Chapter of the International Group for the Psychology of Mathematics Education, Proceedings of the Annual Meeting (13th, Blacksburg, Virginia, October 16–19, 1991). Volumes 1 and 2. Oct. 1991.
   ERIC Microfiche ED352274.
- Urion, David K. "Student Achievement in Small–Group Instruction versus Teacher–Centered Instruction in Mathematics." *Primus.* 1992, 2(3), pp. 257–264.
- von Glaserfeld, Ernst. "An Introduction to Radical Constructivism" in P. Watzlawick. ed. *The Invented Reality.* 1984. pp. 17–40.

von Glaserfeld, Ernst. ed. Radical Constructivism in Mathematics Education. Lawrence Erlbaum Associates,

Hillsdale, NJ, 1991.

- Wallace, Josephine D. "The Concept Map as a Research Tool: Exploring Conceptual Change in Biology." *Journal of Research in Science Teaching.* 1990, 27(10), pp. 1033–1052.
- Watson, Jane. "Research for Teaching." *Australian Mathematics Teacher.* 1991, 47(4), pp. 18–19.
- Wavering, Michael J. What Do Prospective Science Teachers Understand About the Nature of Science? 1990. ERIC Microfiche ED319603.
- Waywood, Andrew. "Journal Writing and Learning Mathematics." *For the Learning of Mathematics.* 1992, 12(2), pp. 34–43.
- Webb, Noreen M., et. al. Mathematical Problem–Solving Processes and Performance: Translation among Symbolic Representations. Feb. 1990. ERIC Microfiche ED344748.
- Wenner, George. "Relationship Between Science Knowledge Levels and Beliefs Toward Science Instruction Held by Preservice Elementary Teachers." *Journal of Science Education and Technology*. 1993, 2(3), pp. 461–468.
- Wheelock, Anne. Crossing the Tracks: How 'Untracking' Can Save America's Schools. 1992. ERIC Microfiche ED353349.
- White, J. "Student Teaching as a Rite of Passage." Anthropology and Education Quarterly. 1989, 89(20), pp. 177–195.
- Wilcox, Sandra K., et. al. Influencing Beginning Teachers' Practice in Mathematics Education: Confronting Constraints of Knowledge, Beliefs, and Context. Feb. 1992. ERIC Microfiche ED343888.
- Wilcox, Sandra K., et. al. "The Role of a Learning Community in Changing Preservice Teachers' Knowledge and Beliefs about Mathematics Education." *For the Learning of Mathematics.* 1991, 11(3), pp. 31–39.
- Williamson, Margaret Eileen. Implementing Metacognitive Processing in the Mathematics Classroom. Aug. 1991. ERIC Microfiche ED347070.
- Wood, T., P. Cobb, and E. Yackel. "Change in Teaching Mathematics: A Case Study." *American Education Research Journal*. 1991, 28(3), pp. 587–616.
- Yackel, E., P. Cobb, T. Wood, G. Wheatley, and G. Merkel. "The Importance of Social Interaction in Children's Construction of Mathematical Knowledge" in T. J. Cooney and C. R. Hirsch. eds. *Teaching and Learning Mathematics in the 1990s.* 1992. pp. 12–21.

- Yager, Robert E. "Science Teacher Education in Four-Year Colleges 1960–1985." Journal of Science Teacher Education. 1991, 2(1), pp. 1–6.
- Yager, Robert E. "Science/Technology/Society as a Major Reform in Science Education: Its Importance for Teacher Education." *Teaching Education.* 1991, 3(2), pp. 91–100.
- Yerushalmy, Michal. "Using Empirical Information in Geometry: Students' and Designers' Expectations." *Journal of Computers in Mathematics and Science Teaching.* 1990, 9(3), pp. 23–37.
- Yore, Larry D. *Middle School Students' Metacognitive Knowledge about Science Reading and Science Text: Objective Assessment, Validation, and Results.* 1992. ERIC Microfiche ED356134.

# APPENDIX E: LIST OF P ARTICIPANTS

Daryl Adams Mankato State University

Sarah Aderhold Probstfield School

Vjendra Agarwal Moorhead State University

Mary Jo Aiken National Council of Teachers of Mathematics

Dan Allen Sauk Rapids Senior High School

Edwin Andersen Southwest Senior High School

Bryan Anderson Bethel College

Carol Anderson Winona State University

Debra Anderson St. Olaf College

Mike Anderson Gustavus Adolphus College

Stefan Anderson Breck School

Gary Anfenson St. Cloud Technical High School

Dorothy Anway College of St. Scholastica

Karen Appeldoorn St. Olaf College Jo Asmussen Bemidji State University

Eleonore Balbach Anderson Open School

Anne Bartel SciMath<sup>MN</sup>

Tim Bates University of Minnesota, Duluth

Dennis Battaglini Winona State University

Jerry Beilby Northwestern College

Andrew Bennett Kansas State University

Steve Benson University of New Hampshire

Alan Beth Winona Senior High School

Jim Bettendorf North Junior High School

Char Bezanson St. Olaf College

Kevin Blanchette St. John's Preparatory School

Sandra Blank Cotter High School Thomas Boates Bemidji State University

Paul Boehlke

Dr. Martin Luther College

Tom Boman University of Minnesota, Duluth

Ed Borchardt Mankato State University

Gary Breitag Bemidji Senior High School

David Bressoud Macalester College

Tricia Brownlee Bethel College

Carolyn Bruels Winona Middle School

Robert Brummond Concordia College

Joel Burgeson University of Minnesota

Joan Cady University of St. Thomas

Stephan Carlson University of Minnesota

Lisa Clemens SciMath<sup>MN</sup>

Marcy Copeland Granite Falls/Clarkfield High School Cyndy Crist Minnesota State Colleges and Universities

Arnie Cutler

Mounds View Senior High School

Sarah Dahl Valley Middle School

Mike Damyanovich SciMath<sup>MN</sup>

George Davis Moorhead State University

Buzz DeLaRosby Two Harbors High School

Barbara DeMaster Gustavus Adolphus College

Nancy Desmond College of St. Benedict

Bruce Dickau College of St. Benedict

Gloria Dimoplon Mankato State University

Linda Distad College of St. Catherine

William Doyle Bethel College

Bruce Drewlow Augsburg College

Gail Earles St. Cloud State University Ruby Eiden St. Joseph Lab School

William Eppright Northwestern College Rhonda Erdmann Willow Creek Junior High School

Charles Ernst St. Cloud State University

Janel Fiksdal St. Mary's College

Fred Finley University of Minnesota

Michael Fiske St. Cloud State University

Fred Foss Winona State University

John Frey Mankato State University

James Gallagher Michigan State University

Eric Gossett Bethel College

Hal Gritzmacher Bemidji State University

George Gross Winona State University

Alice Guckin College of St. Scholastica

Julie A. Guelich Normandale Community College

Shobha Gulati St. John's University Martha Gulner St. Paul School District

Jon Gustafson South St. Paul High School

Melisa Hancock Kansas State University

Lynn Hartshorn University of St. Thomas

Orvald B. Haugsby Concordia College

Pat Hauslein St. Cloud State University

Don Hein Southwest State University

Robert J. Hermann Winona State University

Ann Heuschele Normandale Community College

Jan Hintz St. Cloud State University

Ann Hobbie Brimhall Elementary School

Russell Hobbie University of Minnesota Clark C. Hoffman University of Minnesota, Morris

Julie Hoffner Moorhead Senior High School

Mike Holen Kansas State University

# APPENDIX D: LIST OF P ARTICIPANTS, continued

Don Holman Willmar Community College

Alan Holmes Southwest State University

Robert Holtz Concordia College

Joyce Hummel Kennedy School

Mike Hvidsten Gustavus Adolphus College

Charlotte liams Moorhead State University

Huang Jaifen Mankato State University

Marion Johnson Discovery Elementary School

Susan Johnson Northwestern College

Vivian Johnson Augsburg College

Cindy Johnson-Groh College of St. Scholastica Lewis Jones Mankato State University

Clayton Keller University of Minnesota - Duluth

Ken Kelsey St. Cloud State University John Kemper University of St. Thomas

Keith Kennedy St. Cloud State University

Barb Kepner Sartell Intermediate School

Caroline Khyl University of St. Thomas

Roger Klockziem Dr. Martin Luther College

Bev Kochmann St. Cloud State University

John Koser Wayzata Senior High School

Judy Kuechle University of Minnesota, Morris

Linda Lamwers St. Cloud State University

Pam Landers MN Office of Environmental Education

Kil Lee Mankato State University

Mary Ann Lee Mankato State University

Jean Leicester Winona State University

Larry Luck Anoka-Ramsey Community

#### College

Kenneth Lundberg Bemidji State University

Kathleen Lundgren MN Department of Children, Families and Learning

Carol Marxen University of Minnesota, Morris

Dennis Mathiason Moorhead State University

Kathleen Maury Mankato State University

Robert McClure St. Mary's College

Jim McCracken Bemidji State University

Jeff McLean University of St. Thomas

Richard Mesenburg MN Department of Children, Families and Learning Jeff Miller St. Peter High School

Joland Mohr Southwest State University

Ted Molitor University of St. Thomas

Steven Moseman Brimhall Elementary Bruce Munson University of Minnesota, Duluth

Dennis Nielsen Winona State University

Don Nitti Cotter High School

Michael O'Reilly University of Minnesota, Morris

Duane Orr Mankato State University

Arnold Ostebee St. Olaf College

Dale Pearson Highland Park Senior High School

Roger Peckover St. Mary's College

David Pelzl Dr. Martin Luther College

Laurie Peterman SciMath<sup>MN</sup>

Richard Peterson Bethel College

Jim Pierce Mankato State University

James Poff St. John's University

Thomas R. Post University of Minnesota Jeff Pribyl Mankato State University

Betty Punzac St. Joseph's Lab School

Cheryl Quinn Winona State University

Jon Quistgaard Bemidji State University

Becky Raimann Inver Hills Community College

Helen Rallis University of Minnesota, Duluth

Glen Richgels Bemidji State University

Judy Rohde John Glenn Middle School

Bruce Romanish St. Cloud State University

Cris Roosenraad Carleton College

Susan Roosenraad Northfield High School

Joe Rossow Marcy Open School

Ted Rowe Southwest State University

Carolyn Scheibelhut

Concordia College

Pat Schneider Sartell Intermediate School

Cynthia J. Schobel Moorhead State University

Carl Schoenbeck Concordia College

Gordon Schrank St. Cloud State University

Duane Sea Bemidji State University

Donald Sellke Concordia College

Melissa Shepard University of St. Thomas

Joe Shields St. Mary's College

Mary Shimabukuro Moorhead State University

Gail Shroyer Kansas State University

Patty Simpson St. Cloud State University

Sally Sloan Winona State University

Leone Snyder Northwestern College

Sharon Stenglein

## APPENDIX E: LIST OF P ARTICIPANTS, continued

MN Department of Children, Families and Learning

Mark Stensvold College of St. Scholastica

Judy Strong Moorhead State University

David Stueber Concordia College

Ross Taylor Macalester College

Phil Tennison St. Cloud State University

Carol Theisen St. John's University/ St. Benedict's University

Alice Thomas St. Cloud State University

Mary E. Thompson College of St. Catherine

William K. Tomhave Concordia College

Tom Tommet University of St. Thomas

John Truedson Bemidji State University

Walter Ullrich St. Cloud State University

Rebekah Valdivia

Augsburg College

Nancy Gerdin Vall Bethel College

Joseph Van Wie Southwest State University

Tim Velner Nettleton Elementary School

Ted Vessey St. Olaf College

Kenneth Vos College of St. Catherine

Joe Wacker Minnetonka Middle School

Martha Wallace St. Olaf College

Ivan Watkins St. Cloud State University

Jim Whitney SciMath<sup>MN</sup>

Cathy Wick St. Cloud State University

Mary Wiest Mankato State University

Loren Wiger Marshall Junior High School

Denise Wilbur

University of St. Thomas

Dale Williams St. Cloud State University

Mary Ann Wolf New Ulm Area Catholic Schools

Vernon Wolff Moorhead State University

Janet Woodard St. Cloud State University



643 Capitol Square 550 Cedar Street Saint Paul, MN 55101 612/ 296-4058 612/ 297-7340 FAX http://www.informns.k12.mn.us/scimathmn.html